

Received April 2, 2022, accepted April 15, 2022, date of publication April 26, 2022, date of current version May 10, 2022. Digital Object Identifier 10.1109/ACCESS.2022.3170547

Performance Improvement of Multiobjective **Optimal Power Flow-Based Renewable Energy Sources Using Intelligent Algorithm**

TRUONG HOANG BAO HUY¹⁰, TRI PHUOC NGUYEN¹⁰², NURSYARIZAL MOHD NOR¹⁰³, **IRRAIVAN ELAMVAZUTHI**¹⁰³, (Member, IEEE), TAIB IBRAHIM³, AND DIEU NGOC VO^{102,4} ¹Institute of Engineering and Technology, Thu Dau Mot University, Thu Dau Mot, Binh Duong 75109, Vietnam

²Department of Power Systems, Ho Chi Minh City University of Technology (HCMUT), Ho Chi Minh City 72506, Vietnam

³Department of Electrical and Electronics Engineering, Universiti Teknologi PETRONAS, Seri Iskandar, Perak 32610, Malaysia

⁴Vietnam National University Ho Chi Minh City (VNU-HCM), Thu Duc, Ho Chi Minh City 71308, Vietnam

Corresponding author: Nursyarizal Mohd Nor (nursyarizal_mnor@utp.edu.my)

This work was supported by the Ministry of Higher Education (MOHE) through the Fundamental Research Grant Scheme (FRGS) and the Universiti Teknologi PETRONAS under Grant FRGS/1/2018/TK07/UTP/02/3.

ABSTRACT Producing energy from a variety of sources in a power system requires an optimal schedule to operate the power grids economically and efficiently. Nowadays, power grids might include thermal generators and renewable energy sources (RES). The integration of RES adds complexity to the optimal power flow problem due to intermittence and uncertainty. The study suggests a Multi-Objective Search Group Algorithm (MOSGA) to deal with multi-objective optimal power flow integrated with a stochastic wind and solar powers (MOOPF-WS) problem. Weibull and lognormal probability distribution functions (PDFs) are respectively adopted to describe uncertainties in wind speed and solar irradiance. The MOSGA incorporates crowding distance strategies, fast non-dominated sorting, and an archive selection mechanism to define and preserve the best non-dominated solutions. The total cost, real power loss, and emission were defined as the objectives for the MOOPF-WS problem. In the economic aspect, formulated cost modelling includes both overestimation and underestimation situations related to wind and solar power prediction. Further, uncertainty in load demand is represented by a normal PDF and is considered as a special study case due to its novelty. The effectiveness of MOSGA was validated on the IEEE 30-bus and 57-bus systems considering various combinations of objective functions as well as different loading scenarios. Its performance was comprehensively compared with the other three well-regarded multi-objective optimization algorithms including NSGA-II, MOALO, and MOGOA in terms of Spread metric, Hypervolume metric, and the best compromise solutions for all scenarios. The comparisons showed MOSGA was capable of obtaining well-distributed Pareto fronts and producing better quality solutions compared to the others in all tested scenarios. In addition, MOSGA also obtained better solution quality than significant research in the literature for all the comparable cases. These show the superiority of the MOSGA in dealing with the MOOPF-WS problem.

INDEX TERMS Multi-objective search group algorithm, multi-objective optimal power flow, renewable energy sources, wind power, solar power.

I. INTRODUCTION

The optimal power flow (OPF) was developed by Carpentier [1], and the OPF problem has become a significant function in the production and operation of modern energy

The associate editor coordinating the review of this manuscript and approving it for publication was Akshay Kumar Saha¹⁰.

systems [2]. Generation cost minimization is the primary objective of traditional OPF. However, environmental concerns and technical issues have led to the consideration of different objectives, including emissions, real power loss, voltage stability, and voltage profile. Thus, multi-objective OPF (MOOPF) was formulated to optimize a group of objectives simultaneously in power systems. In general,

the MOOPF problem considers fossil fuel-powered thermal generators. Renewable energy sources (RES), which are environmentally friendly and sustainable, provide a better solution, reducing the environmental impact and limiting fossil fuel consumption. Studies of OPF integrating RES have become necessary due to their increasing application in power systems.

In the beginning stage of problem exploration, the MOOPF problem was handled by several conventional methods like gradient-based method [3], linear programming [4], quadratic programming [5], Newton-based method [6], and interior point method [7]. Despite having good convergence characteristics and effectively handling inequality constraints, such conventional methods cannot ensure global optimum result [8]. In addition, conventional approaches are designed specifically for each type of MOOPF problem. In other words, each variant of the MOOPF problem is related to a certain solution approach. Further, these approaches face many difficulties when coping with the attributes of the problem such as integer and discrete decision variables; nondifferentiable objective function; and large range of search space. Moreover, the application of these methods to solving the MOOPF problem is very complex [9].

In the later stage of exploration, with the advent of a new field of computation known as soft computing, a large number of optimization algorithms belonging to the metaheuristic class have been proposed. The main advantages of such methods are that they can deal with large-scale search spaces and are less dependent on problem characteristics. Moreover, these algorithms are capable of estimating multiple points in the search domain simultaneously due to their population-based nature. Therefore, metaheuristic methods are more effective in finding optimal solutions for OPF and MOOPF problems in comparison with conventional methods [9]. In the literature, the MOOPF problem has been studied extensively, and many powerful algorithms have been applied with remarkable results. In an attempt to deal with the MOOPF problem, Bouchekara et al. [10] proposed improved colliding bodies optimization (ICBO). The performance of ICBO was improved through iterations with increasing numbers of colliding bodies. Chaib et al. [11] studied a backtracking search optimization algorithm (BSA) considering several objective functions: emissions, voltage profile, voltage stability, and fuel costs. In [12], the MOOPF problem was formulated as multi-objective problems by assigning weighting factors for power loss, voltage stability, deviation of voltage, and emissions. A differential search algorithm (DSA) was then proposed to deal with the MOOPF problem, and a moth swarm algorithm (MSA) was also applied, with fourteen different cases based on the weighting factor approach [13]. Additionally, Biswas et al. [14] applied differential evolution (DE) to handle the MOOPF problem with numerous case studies considering different objectives. The authors dealt with the operational constraints of the system by using a combination of self-adaptive penalty and superiority of feasible solution as constraint handling methods. A multi-objective Shaheen *et al.* [15] for the MOOPF problem on 118-bus and 57-bus systems. Pulluri *et al.* [16] presented a mixed crossover operator and self-adaptive strategy to improve the DE for the MOOPF problem. In [17], the authors proposed three different strategies to improve the strength Pareto evolutionary algorithm (ISPEA2). The outcomes showed that these strategies enhanced the uniformity and distribution of Pareto fronts. Recently, the multi-objective dimension-based firefly algorithm (MODFA) was suggested by Chen *et al.* [18] for the MOOPF problem with nine cases in three test systems. Fuzzy affiliation was also used to extract the best compromise solution. Warid *et al.* [9] introduced a new method by combining a Jaya algorithm with a quasi-oppositional (QO) strategy. The new algorithm, QOMJaya, was implemented for the MOOPF problem.

differential evolution (MDE) technique was implemented by

In the past decade, research studies mainly focused on handling the OPF and MOOPF problems with thermal generators. Recently, the OPF problem integrated with RES has attracted the attention of researchers due to the increasing penetration of RESs into power systems. Henerica et al. [19] proposed an OPF solution for an off-grid hybrid dieselsolar PV-battery system. In [20], the MBFA technique was applied to handle the OPF framework considering hydrothermal-wind (HTW) systems. Jabr et al. [21] introduced into the OPF a stochastic model of wind power, forecasting power with a probability or relative frequency histogram. To address the same problem, Mishra et al. [22] developed the bacterial foraging algorithm (BFA) for the OPF to consider wind generation. Stochastic wind speed and power distribution were defined using the Weibull distribution. In [23], the authors introduced a wind-generated cost model that included the opportunity cost of excesses and shortfalls in wind power. The proposed method was implemented using the IEEE New England system. In [24], a modified bacteria foraging algorithm (MBFA) was applied to simulate uncertainty in wind energy generation in the OPF framework. In [25], the MBFA was implemented for the OPF problem considering combined thermal-wind generation. Moreover, the voltage security aspect was improved using shunt FACTS (STATCOM) devices to compensate for reactive power. In [26], the author formulated the OPF problem for a hybrid thermal-wind-solar-storage system. A two-point estimating method and the GA method were implemented to handle the proposed OPF strategy. In [27], the authors solved a security-constrained OPF with wind plant by using a fuzzy adaptive artificial physics algorithm. Roy and Jadhav [28] studied the OPF with wind generation by way of an Improved artificial bee colony (ABC) method. In [29], a hybrid algorithm (HA) was applied for the OPF problem with thermal-wind generators; a modified BFA technique and mutation techniques of the genetic algorithm (GA) were combined to develop the HA method. A technique based on the DE was suggested in [30] for the OPF problem considering solar and wind powers. Some recent applications of metaheuristic methods for OPF problem in

presence of RES such as constrained multi-objective population extremal optimization (CMOPEO) [31], a combination of fitness–distance balance and adaptive guided differential evolution (FDBAGDE) [32], levy interior search algorithm (LISA) [33], grey wolf optimization (GWO) [34], barnacles mating optimizer (BMO) [35], differential evolutionary particle swarm optimization (DEEPSO) [36], and Levy coyote optimization algorithm (LCOA) [37].

A review of the literature has shown that the research on OPF incorporating RESs is indeed encouraging. Although some attempts have been made to address the OPF with the incorporation of RES into power grids, previous research typically only investigated the problem with the singleobjective function of optimal generation cost [21]-[27] or the two-objective function of generation cost and emission [28]–[37]. In addition to economic and environmental factors, technical aspect also needs to be considered when examining the effect of RES on power grids. For this regard, generation cost, real power loss, and emission should be optimized simultaneously in the MOOPF problem. Further, from various combinations of objective functions under different scenario studies, the conflicts between objective functions can be judged. To our knowledge, solving the MOOPF problem with the stochastic RESs using multi-objective algorithms have not been documented in the literature. In addition, many research [38]-[43] considered only solar power or wind power as a renewable source. Very few research [30]-[36] examined both stochastic wind and solar power when solving the MOOPF problem. Further, most of the previous research considered only the invariable load case, which did not reflect the characteristics of the actual load demand. Further study is therefore needed to investigate the MOOPF problem in a system comprising thermal, wind, and solar power generation with the variable loading condition; this constitutes significant motivation for proposing a new technique to determine trade-off solutions for the MOOPF problem considering stochastic wind and solar power along with uncertain load scenario.

The search group algorithm (SGA) is a robust optimization method proposed in [44]. Notably, SGA has the advantage of striking a good balance between exploitation and exploration, providing powerful searchability for finding the optimum solution. Several recent studies have been done to verify the SGA applicability for various optimization problems such as truss optimal voltage regulation in power systems [45], automatic generation control [46], networked control system [47], steel frames optimization [48], and structure optimization [49]. Moreover, SGA has not been applied to deal with the MOOPF problem with the stochastic RESs.

Inspired by the above motivations, this paper aims to suggest a novel multi-objective SGA (MOSGA) for MOOPF incorporating the stochastic wind and solar power (MOOPF-WS) problem. Three approaches are combined into the original SGA to create the proposed MOSGA. The first one named the fast non-dominated sorting technique is applied for determining different non-dominated fronts in a population. The second one named crowding-distance calculation is to preserve the diversity of non-dominated solutions in a specific front. The last one named Pareto archive selection is used to store non-dominated solutions in an archive. The archive is updated via a selection mechanism after each iteration to avoid erroneous rejection of potential candidate solutions during the optimization process. Further, a decisionmaking process is implemented to obtain the best compromise solution. The proposed MOSGA is implemented to find the optimal operation points of the thermal, wind, and solar generators with different case studies of objective functions.

This study contributes to the literature as follows:

- The MOOPF-WS problem was formulated as a constrained multi-objective optimization problem considering the scheduling of thermal power generation and stochastic wind and solar power generation, whereas the references [38]-[43] considered either solar power or wind power as a renewable source. Three vital objective functions of total operational cost, power loss, and emission were concurrently optimized, whereas only a single objective or two objectives were optimized in the references [21]-[37]. The control variables of the problem included most of the important parameters related to the operation of a power transmission network such as the active power output of thermal generators, the active power output of wind and solar generators, voltage magnitudes at generation buses, settings of transformer tap, and reactive power output of shunt VAR compensators. Most of the previous relevant research in [30]-[36] examined the problem with a fewer number of control variables that did not reflect the actual operating condition of the transmission network.
- A new MOSGA was developed to handle the MOOPF-WS problem. Three improvement techniques are integrated into the search mechanism of the original SGA to convert it into a true multi-objective optimization algorithm. The development of MOSGA aims to provision Pareto optimal front and respective trade-offs for conflicting objectives. Meanwhile, previous significant research [30]-[36] used a weighted factors method for transforming from a multi-objective function into a weighted sum function and applied a single-objective optimization algorithm as the optimizer. This workaround only offered a unique optimal solution after a run, which did not reflect the degree of conflict between objectives as well as not provide multiple options in terms of various operating solutions for decision-makers.
- A practical case study relative to uncertainty in load demand was examined to the MOOPF-WS problem where a normal probability distribution function was adopted to describe the variable situation of load demand. Notably, this case study has been investigated for the first time in the MOOPF-WS problem.
- The effectiveness of MOSGA was tested on adapted 30-bus and 57-bus systems incorporating RES with the

consideration of various scenarios. Specifically, in the two first scenarios of each network, the MOOPF-WS problem was investigated in the fixed load situation with two simultaneously optimized objectives, whereas the third scenario was studied in the same load condition for the simultaneous optimization of three objectives. For the last scenario of each network, the problem was solved in the variable load condition while minimizing three objectives concurrently.

• The performance of MOSGA was comprehensively compared with three other multi-objective optimization algorithms including NSGA-II, MOALO, and MOGOA for all scenarios. Comparisons were made in terms of Spread metric, Hypervolume metric as well as the best compromise solution. In all scenarios, statistical results showed the MOSGA achieved better convergence and distribution of Pareto optimal solutions than other multiobjective algorithms. Further, when comparing with previous significant research [30]–[34], MOSGA acquired better solution quality in all the comparable cases.

Section 2 introduces mathematical models for the generation costs of thermal, wind, and solar generators. Section 3 formulates the MOOPF-WS problem. Section 4 outlines the proposed MOSGA and its implementation for the MOOPF-WS problem. Section 5 details the simulation outcomes, and Section 6 presents the conclusions.

II. MATHEMATICAL MODELS

A. COST OF THERMAL GENERATION

The generation cost of thermal generators can be modeled as follows:

$$C_T(P_{TG}) = \sum_{i=1}^{N_{TG}} (a_i + b_i P_{TG,i} + c_i P_{TG,i}^2)$$
(1)

where a_i , b_i , and c_i are the cost constants of the *i*th thermal generator that generates output power $P_{TG,i}$, and N_{TG} denotes the total number of thermal generators.

B. COST OF WIND GENERATION

Being a renewable energy source, wind power has inherent uncertainty related to variable wind power. Due to insufficiency and unavailability, wind turbines cannot effectively provide schedule power as expected in some cases. As a spinning reserve control unit, the independent system operator (ISO) is in charge of handling the deficit amount. Such a situation is referred to as the overestimation of power from renewable sources and should be included in the generation cost because of keeping the spinning reserve. In the contrast, if available power produced by wind turbines is greater than schedule power, the ISO should pay a penalty cost for the surplus amount. This situation is known as underestimation. Therefore, the cost of wind power generation relates to three components of direct cost, reserve cost, and penalty cost [30]. The direct cost of a wind power plant can be expressed as [30]:

$$C_{w,j}(P_{ws,j}) = g_j P_{ws,j} \tag{2}$$

where g_j denotes the direct cost coefficient and $P_{ws,j}$ the scheduled power for the *j*th wind power plant.

As above mentioned, there might be another two costs caused by overestimation and underestimation situations. Precisely, the overestimation situation will lead to the reserve cost, whereas the underestimation one will lead to the penal cost. The wind power plant's reserve cost can be determined as [30]:

$$C_{Rw,j}(P_{ws,j} - P_{wav,j}) = K_{Rw,j}(P_{ws,j} - P_{wav,j})$$

= $K_{Rw,j} \int_{0}^{P_{ws,j}} (P_{ws,j} - p_{w,j}) f_w(p_{w,j}) dp_{w,j}$
(3)

where $P_{wav,j}$, $K_{Rw,j}$, and $f_w(p_{w,j})$ represent the actual available power, reserve cost constant, and probability density function for the *j*th wind power plant, respectively.

The wind power plant's penalty cost can be determined as [30]:

$$C_{Pw,j}(P_{wav,j} - P_{ws,j}) = K_{Pw,j}(P_{wav,j} - P_{ws,j})$$

= $K_{Pw,j} \int_{P_{ws,j}}^{P_{wr,j}} (p_{w,j} - P_{ws,j}) f_w(p_{w,j}) dp_{w,j}$
(4)

where $P_{wr,j}$ and $K_{Pw,j}$ denote rated output power and the penalty cost constant for the j^{th} wind power plant, respectively.

The wind speed distribution is defined using the Weibull probability density function (PDF) with two parameters [25], [50]. Hence, the probability of wind speed (v) is expressed as:

$$f_{\nu}(\nu) = \left(\frac{k}{c}\right) \left(\frac{\nu}{c}\right)^{k-1} e^{-\left(\frac{\nu}{c}\right)^{k}}; \text{ for } 0 \le \nu \le \infty$$
(5)

where k is the shape coefficient and c is the scale coefficient for the PDF.

The mean of the Weibull distribution can be stated as:

$$M_{wbl} = c \times \Gamma(1 + k^{-1})$$

$$\Gamma(x) = \int_0^\infty e^{-t} t^{x-1} dt$$
(6)

The wind turbine power output can be defined as:

$$p_{w}(v) = \begin{cases} 0 & v < v_{in} \text{ and } v > v_{out} \\ p_{wr} \left(\frac{v - v_{in}}{v_r - v_{in}}\right) & v_{in} \le v \le v_r \\ p_{wr} & v_r < v \le v_{out} \end{cases}$$
(7)

where p_{wr} represents the rated output power of the wind turbine; v_{in} , v_r , and v_{out} respectively denote the cut-in speed, rated speed, and cut-out speed of wind turbine.

According to Eq. (7), in a couple of zones of wind speeds, the variable wind power is discrete. Specifically, the wind power is zero for wind speed v smaller than v_{in} or for v greater than v_{out} . The wind power is equal to the rated value when v is between v_r and v_{out} . Probabilities for these discrete zones can be calculated based on Weibull cumulative distribution function (cdf) as follows [51]:

$$f_{w}(p_{w})\{p_{w} = 0\}$$

$$= cdf(v_{in}) + (1 - cdf(v_{out}))$$

$$= 1 - \exp\left[-\left(\frac{v_{in}}{c}\right)^{k}\right] + \exp\left[-\left(\frac{v_{out}}{c}\right)^{k}\right] \qquad (8)$$

$$f_{w}(p_{w})\{p_{w} = p_{wr}\}$$

$$= cdf(v_{out}) + (1 - cdf(v_r))$$
$$= \exp\left[-\left(\frac{v_r}{c}\right)^k\right] - \exp\left[-\left(\frac{v_{out}}{c}\right)^k\right]$$
(9)

For wind speed being between v_{in} and v_r , the wind power is a continuous variable. The continuous zone is related to the probability as follows:

$$f_w(p_w) = \frac{k(v_r - v_{in})}{c^k \times p_{wr}} \left[v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in}) \right]^{k-1} \\ \times \exp\left[-\left(\frac{v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in})}{c}\right)^k \right]$$
(10)

Given the probabilities of available wind power at different operating zones, Eqs. (3) and (4) can be expanded to include the PDF of wind power as in Eqs. (11) and (12), as shown at the bottom of the page, respectively.

Remark 1: The process of handling the stochastic wind power can be summarized as follows: Firstly, the wind speed distribution according to the Weibull probability density function (PDF) [41], [52]–[54] is established using Eqs. (5) and (6). With the known parameters of Weibull shape (k) and scale (c), the Monte-Carlo method is adopted for generating the wind speed frequency distribution and Weibull fitting. Next, based on the power output function

of a turbine in Eq. (7) and the mathematical equations of Weibull distribution, the available wind power probabilities corresponding to different wind speed zones can be defined by Eqs. (8) - (10). Lastly, thanks to the probability functions for available wind power estimation at different operating zones, we can compute the cost of stochastic wind power using Eqs. (11) and (12).

Overall, the generation cost of wind generators can be defined as:

$$C_{WG} = \sum_{j=1}^{N_{WG}} [C_{w,j}(P_{ws,j}) + C_{Rw,j}(P_{ws,j} - P_{wav,j}) + C_{Pw,j}(P_{wav,j} - P_{ws,j})]$$
(13)

where N_{WG} represents the number of wind generators.

C. COST OF SOLAR GENERATION

Besides wind source, solar PV is another well-known renewable source. Generation of a solar PV plant also includes a summation of direct cost, reserve cost, and penalty cost components. The direct cost of the solar PV plant can be modeled as in Eq. (14) [30]:

$$C_{s,k}(P_{ss,k}) = h_k P_{ss,k} \tag{14}$$

where h_k and $P_{ss,k}$ denote the direct cost constant and scheduled power for the k^{th} solar PV plant.

The solar PV plant' reserve cost can be obtained using the below relation [30]:

$$C_{Rs,k}(P_{ss,k} - P_{sav,k})$$

$$= K_{Rs,k}(P_{ss,k} - P_{sav,k})$$

$$= K_{Rs,k} \times f_s(P_{sav,k} < P_{ss,k}) \times [P_{ss,k} - E(P_{sav,k} < P_{ss,k})]$$
(15)

where $P_{sav,k}$ and $K_{Rs,k}$ are the actual available power and the reserve cost constant of the k^{th} solar PV plant, respectively; $f_s(P_{sav,k} < P_{ss,k})$ denotes the occurrence probability of solar power shortage from the schedule power $(P_{ss,k})$; and $E(P_{sav,k} < P_{ss,k})$ signifies the expectation of solar power being less than $P_{ss,k}$.

$$C_{Rw,j}(P_{ws,j} - P_{wav,j}) = K_{Rw,j} \int_{0}^{P_{ws,j}} \left[(P_{ws,j} - p_{w,j}) \times \frac{k(v_r - v_{in})}{c^k \times p_{wr}} \left(v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in}) \right)^{k-1} \times \exp\left(- \left(\frac{v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in})}{c} \right)^k \right) \right] + (P_{ws,j} - 0) \times f_w(p_{w,j}) \{ p_{w,j} = 0 \}$$

$$C_{Pw,j}(P_{wav,j} - P_{ws,j}) = K_{Pw,j} \int_{P_{ws,j}}^{P_{wr,j}} \left[(p_{w,j} - P_{ws,j}) \times \frac{k(v_r - v_{in})}{c^k \times p_{wr}} \left(v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in}) \right)^{k-1} \times \exp\left(- \left(\frac{v_{in} + \frac{p_w}{p_{wr}}(v_r - v_{in})}{c} \right)^k \right) \right]$$

$$+ (p_{wr} - P_{ws,j}) \times f_w(p_{w,j}) \{ p_{w,j} = p_{wr} \}$$

$$(12)$$

VOLUME 10, 2022

The solar PV plant' penalty cost is as follows [30]:

$$C_{Ps,k}(P_{sav,k} - P_{ss,k})$$

$$= K_{Ps,k}(P_{sav,k} - P_{ss,k})$$

$$= K_{Ps,k} \times f_s(P_{sav,k} > P_{ss,k}) \times [E(P_{sav,k} > P_{ss,k}) - P_{ss,k}]$$
(16)

where $K_{Ps,k}$ are the penalty cost constant relative to the k^{th} solar PV plant; $f_s(P_{sav,k} > P_{ss,k})$ denotes the probability that solar power is more than the schedule power $(P_{ss,k})$; and $E(P_{sav,k} > P_{ss,k})$ signifies the expectation of solar power being more than $P_{ss,k}$.

Lognormal PDF is used to represent the solar irradiance (G_s) distribution. Hence, the probability of solar irradiance (G_s) is defined by the following equation [55]:

$$f_G(G_s) = \frac{1}{G_s \sigma \sqrt{2\pi}} \exp\left\{\frac{-(\ln x - \mu)^2}{2\sigma^2}\right\} \text{for}G_s > 0 \quad (17)$$

where μ is the mean and σ is the standard deviation for the PDF.

The mean of the lognormal PDF can be determined as follows:

$$M_{lgn} = \exp\left(\mu + \frac{\sigma^2}{2}\right) \tag{18}$$

The solar irradiance (G_s) to energy conversion for solar PV can be determined as follows [56]:

$$P_{s}(G_{s}) = \begin{cases} P_{sr}\left(\frac{G_{s}^{2}}{G_{std}R_{c}}\right) & \text{for} 0 < G_{s} < R_{c} \\ P_{sr}\left(\frac{G_{s}}{G_{std}}\right) & \text{for} G_{s} \ge R_{c} \end{cases}$$
(19)

where P_{sr} denotes the rated output power of solar PV plant, R_c is the certain irradiance point, and G_{std} denotes solar irradiance in the standard environment.

The reserve cost in Eq. (15) is rewritten as follows [30]:

$$C_{Rs}(P_{ss} - P_{sav}) = K_{Rs}(P_{ss} - P_{sav})$$

= $K_{Rs} \sum_{n=1}^{N^{-}} [P_{ss} - P_{sn-}] \times f_{sn-}$ (20)

where P_{sn-} signifies that available power falls short of scheduled power, f_{sn-} denotes the relative frequency of occurrence of P_{sn-} , and N^- signifies the number of pairs (P_{sn-} , f_{sn-}) created for the PDF.

Similarly, the penalty cost in Eq. (16) is defined as [30]:

$$C_{Ps}(P_{sav} - P_{ss}) = K_{Ps}(P_{sav} - P_{ss})$$

= $K_{Ps} \sum_{n=1}^{N^{+}} [P_{sn+} - P_{ss}] \times f_{sn+}$ (21)

where P_{sn+} signifies that available power is greater than scheduled power, f_{sn+} denotes the relative frequency of occurrence of P_{sn+} , and N^+ signifies the number of pairs (P_{sn+}, f_{sn+}) created for the PDF.

Remark 2: Similarly, the process of handling the stochastic solar power can be summarized as follows: The lognormal

PDF having characteristics as in Eqs. (17) and (18) is used to express the distribution of solar irradiance. Via Monte Carlo simulation, we can obtain the lognormal fitting and frequency distribution of solar irradiance. Next, we calculate the solar irradiance to energy conversion for solar PV plant using Eq. (19). Given the distribution of actual available solar power, the reserve and penalty costs related to stochastic solar power can be computed using Eqs. (20) and (21), respectively.

Overall, the generation cost of solar generators is as follows:

$$C_{SG} = \sum_{k=1}^{N_{SG}} [C_{s,k}(P_{ss,k}) + C_{Rs,k}(P_{ss,k} - P_{sav,k}) + C_{Ps,k}(P_{sav,k} - P_{ss,k})]$$
(22)

where N_{SG} represents the number of solar PV generators.

III. PROBLEM FORMULATION

The MOOPF-WS problem is set out to optimize predefined objectives while meeting several constraints by defining the optimum values of control variables. Therefore, the MOOPF-WS problem may be expressed accordingly:

Minimize:

$$F(u, x) = [F_1(u, x), F_2(u, x), \dots, F_m(u, x)]$$
(23)

Subject to:

$$h(u, x) = 0 \tag{24}$$

$$g(u,x) \le 0 \tag{25}$$

where $F_m(u,x)$ is the *m*th objective function; *u* and *x* are the vectors of control and state variables, respectively; h(u,x) denotes the set of equality constraints and g(u,x) represents the inequality constraints.

A. PREDEFINED OBJECTIVES

1) TOTAL COST MINIMIZATION

The total cost includes generation costs for the thermal, wind, and solar PV generators, which is defined as a summation of Eqs. (1), (13), and (22):

$$F_1 = C_T(P_{TG}) + C_{WG} + C_{SG}$$
(26)

2) POWER LOSS MINIMIZATION

Real power loss can be expressed by the following equation:

$$F_2 = P_{loss} = \sum_{q=1}^{N_l} G_{q(ij)} \left[V_i^2 + V_j^2 - 2V_i V_j \cos(\delta_i - \delta_j) \right]$$
(27)

where N_l denotes the number of transmission lines, $G_{q(ij)}$ denotes the transfer conductance between buses *i* and *j*; V_i and V_j represent the voltage magnitudes at buses *i* and *j*, respectively; δ_i and δ_j represent voltage angles at buses *i* and *j*, respectively.

3) EMISSION MINIMIZATION

Producing electricity using conventional fossil fuels would release harmful emissions into the environment. The total emission caused by thermal generators is defined as follows:

$$F_{3} = \sum_{i=1}^{N_{TG}} \left[(\alpha_{i} + \beta_{i} P_{gi} + \gamma_{i} P_{gi}^{2}) + \omega_{i} e^{(\mu_{i} P_{gi})} \right]$$
(28)

where α_i , β_i , γ_i , ω_i , and μ_i denote the emission constants of the *i*th thermal generator.

B. CONTROL VARIABLES

1) CONTROL VARIABLES

The set of control variables used is as follows:

- *P*_{*TG*,*i*}: real power output of the ith thermal generator, except for the slack bus.
- $P_{WG,i}$: real power output of the ith wind generator.
- $P_{SG,i}$: real power output of the ith solar PV generator.
- $V_{G,i}$: voltage magnitude at the ith generation bus.
- T_i : setting of the ith transformer tap.
- $Q_{C,i}$: reactive power output of the *i*th shunt VAR compensator.

The vector of control variables (*u*) is stated as:

$$u = [P_{TG,2}, \dots, P_{TG,N_{TG}}, P_{WG,1}, \dots, P_{WG,N_{WG}}, P_{SG,1}, \dots, P_{SG,N_{SG}}, V_{G,1}, \dots, V_{G,N_{e}}, T_{1}, \dots, T_{N_{t}}, Q_{1}, \dots, Q_{C,N_{c}}]^{T}$$
(29)

in which N_g denotes the total number of generation buses, N_t denotes the number of transformers, and N_c denotes the number of switchable capacitors.

2) STATE VARIABLES

The set of state variables used is as follows:

- $P_{G,1}$: real power output at the slack generator.
- Q_G : reactive power output of generation buses.
- V_L : voltage magnitudes at the load buses.
- S_L : power flow in the transmission lines.

Thus, the vector of state variables (x) is expressed as:

$$x = [P_{G,1}, Q_{G,1}, \dots, Q_{G,N_g}, V_1, \dots, V_{L,N_d}, S_{L,1}, \dots, S_{L,N_l}]^T$$
(30)

in which N_d signifies the number of load buses.

C. SYSTEM CONSTRAINTS

1) EQUALITY CONSTRAINTS

$$P_{G,i} - P_{D,i} = V_i \sum_{j=1}^{N_b} V_j \left[G_{ij} \cos(\delta_{ij}) + B_{ij} \sin(\delta_{ij}) \right];$$

$$i = 1, \dots, N_b$$
(31)

$$Q_{G,i} + Q_{C,i} - Q_{D,i} = V_i \sum_{j=1}^{N_b} V_j \left[G_{ij} \cos(\delta_{ij}) - B_{ij} \sin(\delta_{ij}) \right];$$

 $i = 1, \dots, N_b$
(32)

VOLUME 10, 2022

in which N_b denotes the number of buses; $P_{D,i}$ and $Q_{D,i}$ represent real and reactive power demand at bus *i*, respectively; G_{ij} represents the transfer conductance and B_{ij} the susceptance between bus *i* and bus *j*.

2) INEQUALITY CONSTRAINTS

Generator constraints:

$$P_{TG,i}^{\min} \le P_{TG,i} \le P_{TG,i}^{\max}; i = 1, \dots, N_{TG}$$
 (33)

$$P_{WG,j}^{\min} \le P_{WG,j} \le P_{WG,j}^{\max}; j = 1, \dots, N_{WG}$$
 (34)

$$P_{SG,k}^{\min} \le P_{SG,k} \le P_{SG,k}^{\max}; k = 1, \dots, N_{SG}$$
 (35)

$$Q_{TG,i}^{\min} \le Q_{TG,i} \le Q_{TG,i}^{\max}; i = 1, \dots, N_{TG}$$
(36)

$$Q_{WG,j}^{\min} \le Q_{WG,j} \le Q_{WG,j}^{\max}; j = 1, \dots, N_{WG}$$
 (37)

$$Q_{SG,k}^{\min} \le Q_{SG,k} \le Q_{SG,k}^{\max}; k = 1, \dots, N_{SG}$$
(38)

$$V_{G,i}^{\min} \le V_{G,i} \le V_{G,i}^{\max}; i = 1, \dots, N_g$$
 (39)

Shunt VAR compensator constraints:

$$Q_{C,i}^{\min} \le Q_{C,i} \le Q_{C,i}^{\max}; i = 1, \dots, N_c$$
(40)

Transformer constraints:

$$T_i^{\min} \le T_i \le T_i^{\max}; i = 1, \dots, N_t$$
(41)

Security constraints:

$$V_{L,i}^{\min} \le V_{L,i} \le V_{L,i}^{\max}; i = 1, \dots, N_d$$
 (42)

$$S_{L,i} \le S_{L,i}^{\max}; i = 1, \dots, N_l$$
 (43)

IV. MULTI-OBJECTIVE SEARCH GROUP ALGORITHM

The MOSGA is created by integrating fast non-dominated sorting, crowding distance strategies, and Pareto archive selection into the SGA. The original SGA, the new integration strategies, and the proposed MOSGA are detailed as below.

A. SGA

SGA is a metaheuristic method developed in [44]. An introduction of the basic SGA is described as follows. To start the optimization algorithm, a population \mathbf{P} can be generated randomly as the following equation:

$$P_{ij} = x_{j,\min} + (x_{j,\max} - x_{j,\min})U[0, 1];$$

$$i = 1, \dots, n_{pop}, j = 1, \dots, n$$
(44)

where n_{pop} is the population size and *n* denotes the number of design variables; P_{ij} denotes the *j*th control variable of the *i*th individual of **P**, *U* [0, 1] denotes a random number ranging from 0 to 1; $x_{j,\min}$ and $x_{j,\max}$ respectively denote the lower and upper limits of the *j*th control variable.

The objective function value for each individual is computed after population initialization. A number of good individuals are then extracted from population \mathbf{P} to create a search group \mathbf{R} using a standard tournament selection. Further information on the tournament selection is given in [57]. The search group \mathbf{R} is mutated at every iteration to improve global exploration. Depending on the rank in search group \mathbf{R} , the worst member is chosen for mutation. The aim is to explore

TABLE 1. Pseudocode of SGA.

Algorithm 1: Pseudocode of SGA

- 1: Initialize the initial parameters of the SGA;
- 2: Randomly generate an initial population P using Eq. (44);
- 3: Calculate the objective function values for population P;
- 4: Create the initial search group \mathbf{R}^k selecting n_g individuals from the initial population;
- 5: Replace nmat individuals by new members created as described in Eq. (45);
- 6: Generate the families F_i using Eq. (46);
- 7: Select the new search group according to the rule:
 - Global phase: search group \mathbf{R}^{k+1} is formed by best member of each family;
 - Local phase: search group \mathbf{R}^{k+1} is formed by best n_g individuals from the population.
- 8: Update α^{k+1} accordingly to Eq. (47);
- 9: Make k = k + 1, if $k > Iter_{max}$, go to Step 10, otherwise return to Step 5;
- 10: Solution found: $\mathbf{x}^* = \mathbf{R}_{1,:}$

TABLE 2. Pseudocode of MOSGA

Algorithm 2: Pseudocode of MOSGA

- 1: Initialize the initial parameters of the MOSGA;
- 2: Randomly generate an initial population P using Eq. (44);
- 3: Calculate multiple objective function values for population P;
- 4: Sort initial population P based on fast non-dominated sorting and crowding distance strategies and store them in the Pareto archive;
- 5: Create the initial search group \mathbf{R}^k selecting n_g individuals from the initial population;
- 6: Replace nmat individuals by new members created as described in Eq. (45);
- 7: Generate the families Fi using Eq. (46);
- 8: Store all newly created individuals in the advanced Pareto archive. Combine the current and advanced archives;
- Select the best solutions for new Pareto archive using Pareto archive selection mechanism;
 Select the new search group according to the rule:
 - Global phase: search group R^{k+1} is formed by best member of each family;
 - Local phase: search group \mathbf{R}^{k+1} is formed by best n_g individuals from Pareto archive;
- 11: Update α^{k+1} accordingly to Eq. (47);
- 12: Make k = k + 1, if $k > Iter_{max}$, go to Step 13, otherwise return to Step 6;

13: Solution found: Pareto optimal solutions in the last Pareto archive.

new areas in the search domain. Therefore, new individuals are mutated as follows:

$$x_{j,\text{mut}} = E[R_{:,j}] + t\varepsilon\sigma[R_{:,j}]; j = 1, \dots, n,$$
 (45)

in which $x_{j,\text{mut}}$ denotes the j^{th} control variable of a specified mutated individual, $R_{:,j}$ signifies the j^{th} column of the search group matrix, E denotes the mean value, t is the coefficient that determines the position of a newly created individual, ε denotes a random variable, and σ denotes the standard deviation.

In SGA, a set comprising each search group member and the respectively generated individuals is a family. Therefore, after the search group is created, each member generates a family by perturbation as follows:

$$x_{j,\text{new}} = R_{ij} + \alpha \varepsilon; j = 1, \dots, n, \tag{46}$$

where α denotes the perturbation constant, which is reduced by the iteration:

$$\alpha^{k+1} = b\alpha^k \tag{47}$$

in which b means the coefficient, which is defined by a combination of the linear function.

SGA comprises global and local phases. The principal aim in the global phase is to explore the entire design domain by creating a new search group from the best member of each



FIGURE 1. Crowding distance computation.



FIGURE 2. Pareto archive selection procedure.

family. To exploit the region of the best current individual, new search group is formed from the best n_g individuals from all the families in the local phase. Algorithm 1 in Table 1 describes the SGA's pseudocode.

B. FAST NON-DOMINATED SORTING TECHNIQUE

For each solution *p* of a set *S*, two entities are computed [58]:

- Domination count n_p : the number of solutions that dom
 - inate the solution *p*.
- S_p : the set of solutions dominated by solution p.

The solutions with a zero domination count (n_p) are contained in the first non-dominated rank (P₁). Afterwards, for each solution *p* with zero domination count, the process visits each solution *q* of set S_p and decreases the domination count of each solution *q* by one. If a member *q* has a zero domination count, it is located in a different list *Q*. These members are also placed in the second front (P₂). This process continues with each member of *Q* to determine the third rank (P₃) and all non-dominated ranks.

C. CROWDING DISTANCE COMPUTATION

Crowding distance is computed to estimate the density of solutions around a specific solution i in a specific non-dominated front. The population needs to be sorted by the values of the objective function in ascending order. Then, an infinite distance value for each objective function is assigned to the solutions with minimum and maximum objective values, i.e., the boundary solutions. A distance value is assigned to all other intermediate solutions; this value is the absolute normalized difference between the objective values of two neighboring solutions (i+1 and i-1). This computation is employed for other objectives. The value for crowding distance is computed as the total of the individual distance values for each objective. Before calculating the crowding distance, each objective function is normalized. Fig. 1 shows a schematic of the crowding distance computation.

The crowded-comparison operator (\prec_n) is used based on non-dominated level (r) and crowding distance (d) to define the better solution:

$$i \prec_n j$$
 if $(r_i < r_j)$ or $((r_i = r_j)$ and $(d_i > d_j))$ (48)

According to Eq. (48), the solution that belongs to the better non-dominated level is prioritized; if both solutions are in the same level, the one with the better value for crowding distance is prioritized.

D. PARETO ARCHIVE SELECTION

The Pareto archive is utilized with the best non-dominated solutions to maintain a non-dominated set. In the MOSGA, all generated solutions are placed in the advanced archive. Two archives are obtained (current and advanced), and these are combined. The combined archive exceeds the fixed size (n_{pop}) , and so this archive is truncated using the selection strategy suggested by Deb *et al.* [58] for the next step of the algorithm. Fig. 2 outlines this procedure.

Fast non-dominated sorting is used first to separate the combined archive into non-dominated ranks $(P_1, P_2, ..., P_n)$. The solutions in the first non-dominated front (P_1) are chosen first for the new archive. If the first front (P_1) is smaller than the archive, the remainder of the archive is selected from other non-dominated fronts in rank order $(P_2, ..., P_n)$. The process is implemented until the Pareto archive reaches its limit, assuming that front F_k is the final non-dominated front, after which no further sets can be added. Solutions from the last front (P_k) are chosen in descending order of the crowding distance to select the number of solutions for the new Pareto archive accurately.

E. THE PROPOSED MOSGA

The MOSGA begins the procedure by randomly generating an initial population **P**. Objective values are calculated for all individuals of **P**. Based on the non-dominated ranks of **P**, a tournament selection is later employed to choose n_g best individuals from **P** to form an initial search group **R** for the subsequent two processes (mutation and generation of the families). Afterwards, the advanced archive is yielded, and the current and advanced archives are combined. The selection strategy is implemented to choose n_{pop} best individuals for the new Pareto archive. Finally, a crucial stage of the MOSGA is to select a new search group. This process in both stages is carried out using a tournament selection. The new search group is generated by the best member of each family

F. IMPLEMENTATION OF THE MOSGA FOR THE MOOPF-WS PROBLEM

1) POPULATION INITIALIZATION

To implement the MOSGA for the MOOPF-WS problem, each individual in the population \mathbf{P} represents a vector of control variables as follows:

$$P_{d} = [P_{TG,1}, \dots, P_{TG,N_{TG}}, P_{WG,1}, \dots, P_{WG,N_{WG}}, P_{SG,1}, \dots, P_{SG,N_{SG}}, V_{G,1}, \dots, V_{G,N_{G}}, T_{1}, \dots, T_{N_{t}}, Q_{1}, \dots, Q_{C,N_{C}}]^{T}$$

$$d = 1, \dots, n_{pop}$$
(49)

2) OBJECTIVE FUNCTION VALUE

In the MOOPF-WS problem, each objective function value is calculated as follows:

$$F'_{m} = F_{m} + K_{p}(P_{G,1} - Q_{G,1}^{\lim})^{2} + K_{q} \sum_{i=1}^{N_{G}} (Q_{G,i} - Q_{G,i}^{\lim})^{2} + K_{v} \sum_{i=1}^{N_{d}} (V_{L,i} - V_{L,i}^{\lim})^{2} + K_{s} \sum_{i=1}^{N_{l}} (S_{L,i} - S_{L,i}^{\max})^{2}$$
(50)

in which F_m denotes the m^{th} objective for the MOOPF-WS; K_p, K_q, K_v , and K_s denote the penalty constants. In this study, penalty constants are set to 10^6 .

The limitation values of the state variables are defined according to Eq. (51):

$$x^{\text{lim}} = \begin{cases} x_{\text{max}} & \text{if } x > x_{\text{max}} \\ x_{\text{min}} & \text{if } x < x_{\text{min}} \\ x & \text{otherwise} \end{cases}$$
(51)

where x^{lim} and x denote the limited values and calculated values of Q_G , V_L , and S_L , respectively.

The power flow problem was addressed using the Matpower 6.0 toolbox [59].

3) BEST COMPROMISE SOLUTION

A decision-making method based on a fuzzy membership function is used to define the best compromise solution. The membership function μ_{ij} represents the degree of satisfaction of the *i*th solution for the *j*th objective [60]:

$$\mu_{ij} = \begin{cases} 1 & \text{if } F_{ij} \le \min(F_j) \\ \frac{\max(F_j) - F_{ij}}{\max(F_j) - \min(F_j)} & \text{if } \min(F_j) \le F_{ij} \le \max(F_j) \\ 0 & \text{if } F_{ij} \ge \max(F_j) \end{cases}$$

(52)



FIGURE 3. The modified IEEE 30-bus system incorporating RES.

where $\min(F_j)$ and $\max(F_j)$ are the minimum and maximum values of the *j*th objective, respectively. The normalized membership function of the *i*th solution can be defined as:

$$\mu_{i} = \frac{\sum_{j=1}^{n_{obj}} \mu_{ij}}{\sum_{i=1}^{n_{obj}} \sum_{j=1}^{n_{obj}} \mu_{ij}}$$
(53)

where n_{pf} denotes the number of objective functions and n_{obj} denotes the number of non-dominated solutions, respectively. The best compromise solution is the one with a maximum value of the normalized membership function.

4) OVERALL PROCEDURE

The MOSGA implementation for the MOOPF-WS is as follows:

Step 1: Define detailed data of the test system and acceptable ranges of control variables;

Step 2: Set the controlling parameters for MOSGA, including population size (n_{pop}) , number of search group members (n_g) , number of mutations (n_{mut}) , perturbation factor (α) , Pareto archive size (N_{Ω}) , maximum number of iterations $(Iter_{max})$, and global iteration ratio (GIR);

Step 3: Create the initial population **P** as in Eq. (44);

Step 4: Execute a power flow calculation using Matpower 6.0 to evaluate objective function values (F_1, F_2, F_3) for each individual of **P**;

Step 5: Sort all the individuals in the initial population **P** using fast non-dominated sorting and crowding distance computations and place them into a Pareto archive;

Step 6: Choose the best n_g solutions from **P** to generate the initial search group \mathbf{R}^k . Set k = 0;

Step 7: Begin the main loop, k = k + 1;

Step 8: Perform the mutation phase for n_{mut} individuals using Eq. (45);

Step 9: Create families (\mathbf{F}_i) for each search group member using Eq. (46);

Step 10: Store all newly generated solutions into an advanced archive. Combine the advanced and current archives;

Step 11: Choose the best N_{Ω} solutions from the combined archive as the new Pareto archive;

Step 12: Select the new search group in two stages:

- Global stage: select the best member of each family to create search group \mathbf{R}^{k+1} ;
- Local stage: select n_g best solutions from the archive to create search group \mathbf{R}^{k+1} .
- *Step 13:* Determine α^{k+1} according to Eq. (47);

Step 14: If $k < Iter_{max}$, return to Step 7; or else, go to Step 15;

Step 15: Solutions achieved: non-dominated solutions in the archive set;

Step 16: Define the best compromise solution based on the defined approach;

G. PERFORMANCE METRICS

1) SPREAD INDICATOR

The diversity of non-dominated solutions of a specified algorithm is provided by the spread (Δ) indicator, which shows the spread of extreme solutions and the distribution of solutions in the generated non-dominated set. This parameter is defined in the following equation [61]:

$$\Delta = \frac{\sum_{i=1}^{m} d(E_i, \Omega) + \sum_{X \in \Omega} \left| d(X, \Omega) - \bar{d} \right|}{\sum_{i=1}^{m} d(E_i, \Omega) + (|\Omega| - m) \bar{d}}$$
(54)

where

$$d(X, \Omega) = \min_{Y \in \Omega, Y \neq X} \|F(X) - F(Y)\|$$
(55)

$$\bar{d} = \frac{1}{|\Omega|} \sum_{X \in \Omega} d(X, \Omega)$$
(56)

where Ω is the generated Pareto optimal front, E_i is the i^{th} extreme solutions in true Pareto front, and *m* represents the number of objective functions. The smaller value of the Δ indicator provides better diversity (i.e., the better extent of spread and distribution) in a non-dominated set.

2) HYPERVOLUME

This metric computes the volume related to members of a non-dominated set in the objective space. The hypervolume (HV) can be computed [62]:

$$HV = \bigcup_{i=1}^{|\Omega|} v_i \tag{57}$$

where v_i is a hypercube for each solution $i \in \Omega$ formed with a reference point W. The HV values after normalization lie within the range [0,1]. It is desirable that the algorithm has a large HV value.

TABLE 3. Different scenarios of the MOOPF-WS problem.

Scenario	Total past	Active power loss	Emission	Load	demand	System
	Total cost			Fixed	Variable	_
Scenario 1	✓	\checkmark		✓		
Scenario 2	\checkmark		\checkmark	\checkmark		IEEE 30-bus
Scenario 3	\checkmark	\checkmark	\checkmark	\checkmark		system
Scenarios 4.1-4.4	\checkmark	\checkmark	\checkmark		\checkmark	-
Scenario 5	\checkmark	\checkmark		√		
Scenario 6	\checkmark		✓	\checkmark		IEEE 57-bus
Scenario 7	\checkmark	\checkmark	\checkmark	\checkmark		system
Scenarios 8.1-8.4	\checkmark	\checkmark	\checkmark		\checkmark	

TABLE 4. Summary of the IEEE 30-bus system incorporating RESs.

System characteristics	Value	Detail
Bus	30	[63]
Branch	41	[63]
Thermal generator (TG1, TG2, TG3)	3	Buses: 1, 2, and 8
Wind plant (WG1, WG2)	2	Buses: 5 and 11
Solar PV plant (SG1)	1	Buses: 13
Shunt VAR compensators	9	Buses: 10, 12, 15, 17,
		20, 21, 23, 24 and 29
Transformers with tap changers	4	Branches: 11, 12, 15
		and 36
Control variables	24	-

 TABLE 5. Cost constant and emission constant of thermal generators for modified IEEE 30-bus system [2].

Generator	TG1	TG2	TG3
Bus	1	2	8
а	0	0	0
b	2	1.75	3.25
с	0.00375	0.0175	0.00834
α	0.04	0.025	0.053
β	-0.055	-0.06	-0.035
γ	0.064	0.056	0.033
ω	0.0002	0.0005	0.002
μ	6.667	3.333	2

V. SIMULATION RESULTS

The MOSGA was implemented to deal with the MOOPF-WS problem on the adapted IEEE 30-bus and 57-bus systems incorporating RES. Table 3 summarizes the scenarios considered in this study. The initial parameters of MOSGA, including n_{pop} , n_g , n_{mut} , α^k , and *GIR*, were chosen as 100, 20, 5, 2, and 0.3, respectively, for all test systems. The optimization process of the MOSGA was independently run thirty times for each scenario. The number of function evaluations (NFEs) was set as 30,000 and 50,000 for the 30-bus and 57-bus systems, respectively. The Pareto archive size was kept at 100 for all trials.

Moreover, MOSGA was compared with multi-objective ant lion optimizer (MOALO) [64], multi-objective grasshopper optimization algorithm (MOGOA) [65], and nondominated sorting genetic algorithm II (NSGA-II) [58]. The population size for all algorithms was fixed at 100. In MOGOA, the maximum and minimum values of decreasing coefficient (c_{max} and c_{min}) were set to 1 and 0.00004, TABLE 6. PDF parameters and cost constants for wind generators for modified IEEE 30-bus system [30].

Wind generator #	WG1 (bus 5)	WG2 (bus 11)
Number of wind turbines	25	20
Rated power (MW)	75	60
Weibull PDF parameters	c = 9, k = 2	c = 10, k = 2
Weibull mean	v = 7.976 m/s	v = 7.976 m/s
Direct cost constant (\$/MW)	1.6	1.75
Reserve cost constant (\$/MW)	3	3
Penalty cost constant (\$/MW)	1.5	1.5

 TABLE 7. PDF parameters and cost constants for solar PV generator for modified IEEE 30-bus system [30].

Solar PV #	SPV (bus 13)
Rated power (MW)	50
Lognormal PDF parameters	$\mu = 6, \sigma = 0.6$
Lognormal mean	$G = 483 \text{ W/m}^2$
Direct cost constant (\$/MW)	1.6
Reserve cost constant (\$/MW)	3
Penalty cost constant (\$/MW)	1.5

respectively. NSGA-II used a mutation probability of 0.5, a crossover probability of 0.9, and distribution indexes for mutation and crossover operators of $\eta_m = 20$ and $\eta_c = 20$, respectively.

A. IEEE 30-BUS SYSTEM

Table 4 summarizes the modified IEEE 30-bus system incorporating the RES used in this study; its topology diagram is displayed in Fig. 3. Of note, locations of RESs were chosen similar to the references [30] to make a fair results comparison. In this context, some conventional generators were replaced by respective renewable generators. Table 5 details the fuel cost and emission factors for thermal generators. Tables 6-7 give PDF parameters and cost constants for wind and solar PV generators. After performing 8,000 Monte-Carlo scenarios, Figs. 4-5 depict wind frequency distributions for two wind farms. Similarly, Fig. 6 displays the frequency distribution of solar irradiance for a solar PV plant. Stochastic power output from the solar PV plant is presented in Fig. 7. For the initial scenario, the system has a total cost of 834.6627 \$/h, a real power loss of 5.7866 MW, and an emission of 0.2694 ton/h.

TABLE 8. Simulation results obtained by multi-objective algorithms for scenarios 1 and 2.

Parameters	Min	Max		Scen	ario 1			Scena	ario 2	
			MOSGA	NSGA-II	MOALO	MOGOA	MOSGA	NSGA-II	MOALO	MOGOA
P_{TG1} (MW)	50	200	87.2602	84.4137	84.6407	86.5867	100.7536	96.8823	100.9380	101.2253
P_{TG2} (MW)	20	80	38.0082	39.7378	50.5884	38.3918	47.5322	49.0439	47.6395	48.8515
P_{WG1} (MW)	0	75	57.0495	58.3995	57.3018	58.2082	45.5621	46.7923	44.0522	49.1663
P_{TG3} (MW)	10	35	26.7210	26.3622	11.5691	26.5124	18.3870	19.6006	15.6313	12.6419
P_{WG2} (MW)	0	60	41.8188	41.5309	44.5543	41.5451	39.3667	39.1580	34.1195	39.0103
P_{SG1} (MW)	0	50	35.8393	36.2655	38.3737	35.5043	36.1707	36.1618	45.7678	37.1107
V _{TG1} (pu)	0.95	1.1	1.0640	1.0420	1.0541	1.0596	1.0722	1.0434	1.0468	1.0505
V _{TG2} (pu)	0.95	1.1	1.0557	1.0315	1.0447	1.0514	1.0574	1.0331	1.0347	1.0425
V _{WG1} (pu)	0.95	1.1	1.0407	1.0126	1.0260	1.0390	1.0341	1.0120	1.0099	1.0277
V _{TG3} (pu)	0.95	1.1	1.0436	1.0176	1.0272	1.0378	1.0386	1.0173	1.0152	1.0198
V _{WG2} (pu)	0.95	1.1	1.0699	1.0779	1.1000	1.0960	1.0486	1.0881	1.0476	1.0327
V _{SG1} (pu)	0.95	1.1	1.0550	1.0622	1.0542	1.0328	1.0310	1.0601	1.0648	1.0465
Q _{C10} (MVar)	0	5	4.2118	2.8393	0.3708	0.0819	1.9717	2.6122	0.9995	0.6896
Q _{C12} (MVar)	0	5	2.2092	3.1181	3.7932	3.7310	1.8951	3.2688	3.0244	2.6883
Q _{C15} (MVar)	0	5	3.8990	3.6860	0.5683	4.9555	1.0475	4.0385	2.5078	1.9591
Q _{C17} (MVar)	0	5	2.5930	3.7523	2.9165	1.1022	3.0748	3.6970	2.4343	3.4270
Q _{C20} (MVar)	0	5	2.9253	3.4639	4.9722	4.6229	4.6955	2.3673	3.4680	0.8015
Q _{C21} (MVar)	0	5	3.2965	3.2540	1.9241	4.4982	2.8604	3.7899	4.4961	1.8996
Q _{C23} (MVar)	0	5	1.6797	3.2887	1.6227	3.8377	2.3548	2.8929	3.4334	1.7955
Q _{C24} (MVar)	0	5	4.9334	4.3268	4.8508	4.4080	3.1577	4.4009	0.2824	5.0000
Q _{C29} (MVar)	0	5	3.8667	2.5364	2.6443	4.8598	3.3686	3.0551	2.1262	3.2380
T ₁₁ (pu)	0.9	1.1	1.0093	0.9728	1.0268	0.9846	1.0360	0.9901	1.0643	1.0638
T ₁₂ (pu)	0.9	1.1	0.9465	0.9832	0.9606	1.0957	0.9515	0.9528	0.9120	0.9128
T ₁₅ (pu)	0.9	1.1	0.9883	0.9880	0.9816	0.9833	0.9954	0.9850	1.0687	0.9883
T ₃₆ (pu)	0.9	1.1	0.9796	0.9595	0.9724	1.0087	0.9750	0.9620	0.9656	0.9422
Total Cost (\$/h)	-	-	799.1777	801.4558	800.1741	800.4808	784.7058	787.2669	788.0518	785.7272
Real Power Loss (MW)	-	-	3.2970	3.3097	3.6280	3.3484	-	-	-	-
Emission (ton/h)	-	-	-	-	-	-	0.2788	0.2382	0.2815	0.2856



FIGURE 4. Wind speed distribution for wind generators #1 at bus 5.

1) SCENARIO 1: SIMULTANEOUS OPTIMIZATION OF TOTAL COST AND ACTIVE POWER LOSS

The MOSGA was executed to optimize the total cost and real power loss concurrently in scenario 1. Fig. 8 portrays the best Pareto optimal front associated with the best HV metric for scenario 1, which shows a trade-off between the two objectives considered. Table 8 reports the simulation results found by MOSGA, NSGA-II, MOALO, and MOGOA, including optimal values of design variables and objective functions for the best compromise solution. Table 8 also indicates that the MOSGA offered a total cost of 799.1777 \$/h and a real power loss of 3.2970 MW for the best compromise solution. Thus, total cost and real power loss reductions for this scenario



800

FIGURE 5. Wind speed distribution for wind generators #2 at bus 11.

were 4.2514% and 43.0223%, respectively, in comparison with the initial scenario. Moreover, it showed that the best compromise solution obtained by MOSGA (799.1777 \$/h and 3.2970 MW) was better than those obtained by NSGA-II (801.4558 \$/h and 3.3097 MW), MOALO (800.1741 \$/h and 3.6280 MW) and MOGOA (800.4808 \$/h and 3.3484 MW). Therefore, MOSGA is capable of finding a better optimal solution for this case study.

2) SCENARIO 2: SIMULTANEOUS OPTIMIZATION OF TOTAL COST AND EMISSION

In this scenario, the MOSGA minimized the total cost and emission concurrently. Fig. 9 demonstrates the Pareto front



FIGURE 6. Solar irradiance distribution for solar PV at bus 13.



FIGURE 7. Available solar PV power (MW) at bus 13.



FIGURE 8. Pareto front obtained by MOSGA for scenario 1.

obtained by MOSGA for this scenario. As stated above, the best Pareto optimal front is the one that corresponds to the best value for the HV metric. The curve in Fig. 9 shows the conflicting nature of total cost and emission due to the equation characteristics of objective functions. Table 8 displays simulation output obtained from the MOSGA, NSGA-II, MOALO, and MOGOA for the best compromise solution. The total cost and emission for the best compromise solution, yielded by MOSGA were 784.7058 \$/h and 0.2788 ton/h, respectively. These correspond to a reduction in total cost of 5.9853% and an increase in emissions of 3.4754%, as compared to the initial scenario. Compared with NSGA-II,



FIGURE 9. Pareto front obtained by MOSGA for scenario 2.



FIGURE 10. Pareto front obtained by MOSGA for scenario 3.



FIGURE 11. Load bus voltage profiles for scenarios 1, 2, and 3.

MOSGA acquired the best compromise solution with lower total cost and higher emission. For a comprehensive comparison, MOSGA not only found a better result than MOALO and MOGOA in both objectives, but it also obtained the best value for the total cost objective among other algorithms for best compromise solution. Hence, MOSGA can provide a good-quality best compromise solution for scenario 2.

To further investigate its performance, MOSGA was also compared with published results in the literature for scenario 2. Table 9 shows comparative results of MOSGA and other prominent algorithms in the previous

scenario 3.



FIGURE 12. Representation of uncertainty in load by a normal distribution with $\mu_1 = 70$ and $\sigma_1 = 10$.

 TABLE 9. Comparative results of MOSGA and previous studies for scenario 2.

Algorithms	Total cost	Emission
MOSGA	784.7058	0.2788
SHADE-SF [30]	792.516	0.891
CMOPEO [31]	804.5355	0.5167
CNSGA-II [31]	805.5781	0.6018
FDBGADE [32]	786.1897	0.9216
ISA [33]	793.021	0.813
LISA strategy I [33]	808.307	0.729
LISA strategy II [33]	807.162	0.642
GA [34]	787.7	1.36
PSO [34]	791.88	0.98
CSA [34]	793.13	0.92
ABC [34]	793.46	0.89
GWO [34]	792.73	0.86

studies, including success history based adaptive differential evolutionary with superiority of feasible solutions (SHADE-SF) [30], CMOPEO [31], constrained NSGA-II (CNSGA-II) [31], FDBAGDE [32], ISA [33], LISA strategy I [33], LISA strategy II [33], GA [34], PSO [34], crow search algorithms (CSA) [34], ABC [34], and GWO [34]. As for scenario 2, MOSGA yielded the best minimum value of total cost and emission objectives in best compromise solution among published results. Hence, MOSGA can provide a good-quality best compromise solution for scenario 2.

3) SCENARIO 3: SIMULTANEOUS OPTIMIZATION OF TOTAL COST, ACTIVE POWER LOSS, AND EMISSION

The total cost, real power loss, and emission were minimized concurrently by way of the MOSGA in this scenario. Fig. 10 portrays the Pareto optimal front found by the MOSGA, indicating the relations between the three objectives. The best compromise solution was also extracted through the decision-making method. Table 10 represents simulation outcomes for the best compromise solutions yielded by MOSGA, NSGA-II, MOALO, and MOGOA. Table 10 also shows that the best compromise solution provided a total cost of 807.6874 \$/h, a power loss of 3.0442 MW, and emission of 0.1342 ton/h, corresponding to a total cost reduction of

33	92	

4

Parameters	MOSGA	NSGA-II	MOALO	MOGOA
P _{TG1} (MW)	77.8160	73.5283	75.6901	79.5689
P_{TG2} (MW)	41.4778	47.7872	42.4501	39.6572
P_{WG1} (MW)	60.4941	56.8714	65.2252	61.6539
P_{TG3} (MW)	29.0592	29.7145	11.9243	31.2635
P_{WG2} (MW)	41.5488	41.5110	51.0904	39.4725
P_{SG1} (MW)	36.0483	37.1260	40.1614	34.8559
V _{TG1} (pu)	1.0621	1.0588	1.0564	1.0684
V _{TG2} (pu)	1.0534	1.0461	1.0457	1.0562
V _{WG1} (pu)	1.0346	1.0222	1.0307	1.0459
V _{TG3} (pu)	1.0409	1.0297	1.0313	1.0379
V _{WG2} (pu)	1.0605	1.0508	1.0589	1.0811
V _{SG1} (pu)	1.0663	1.0504	1.0274	1.0476
Q _{C10} (MVar)	3.2258	1.7242	4.2364	4.9770
Q _{C12} (MVar)	2.1065	2.7411	0.4017	1.4943
Q _{C15} (MVar)	2.3569	2.0901	3.2604	5.0000
Q _{C17} (MVar)	3.2783	3.2354	1.7343	4.8643
Q _{C20} (MVar)	1.9136	3.4792	4.8988	1.0396
Q _{C21} (MVar)	3.4459	1.9895	1.9876	4.9496
Q _{C23} (MVar)	2.6009	2.7361	4.8610	1.4816
Q _{C24} (MVar)	4.5637	2.2284	4.5499	4.9061
Q _{C29} (MVar)	2.0493	2.3987	0.9339	2.9942
T ₁₁ (pu)	1.0046	0.9834	0.9638	1.0007
T ₁₂ (pu)	1.0301	0.9673	1.1000	1.0886
T ₁₅ (pu)	0.9883	0.9948	1.0424	0.9753
T ₃₆ (pu)	0.9741	0.9681	0.9595	0.9997
Total Cost (\$/h)	807.6874	809.0396	816.1904	808.4487
Real Power Loss	3.0442	3.1384	3.1414	3.0718
(MW)				
Emission (ton/h)	0.1342	0.1233	0.1312	0.1395

TABLE 10. Simulation results obtained by multi-objective algorithms for

TABLE 11. Scenarios of loading and corresponding probabilities.

Items	Scenario of loading						
	Scenario 4.1	Scenario 4.2	Scenario 4.3	Scenario 4.4			
Percent loading (P_l^{mean})	54.7486	65.4013	74.5986	85.2512			
Probability of scenario (δ_{sce})	0.1587	0.3413	0.3413	0.1587			

3.2319%, a real power loss reduction of 47.3919%, and an emission reduction of 50.1767%, respectively, in comparison with the initial scenario. It can be seen that MOSGA dominated MOGOA in all three objectives in the best compromise solution. Moreover, MOSGA outperformed NSGA-II and MOALO in two out of three objectives. Therefore, MOSGA had great potential in finding a better compromise solution.

According to the results obtained from this scenario, all three objective functions indicate a remarkable improvement using the MOSGA. Precisely, all the objectives were met simultaneously to the maximum extent, an optimal result for the best compromise solution. Moreover, an essential constraint in the MOOPF-WS problem is that load bus voltage must be maintained within a specific range; Fig. 11 shows voltage profiles of load buses for scenarios 1-3, and it can be seen that the voltages are all within specified boundaries.

TABLE 12.	Simulation results obtained by proposed MOSGA algorithm for
loading sce	enarios 4.1-4.4.

Parameters	Scenario	Scenario	Scenario	Scenario		
	4.2	4.3	4.3	4.4		
P_{TG1} (MW)	50.0199	50.0416	52.0470	61.1147		
P_{TG2} (MW)	31.4462	22.5247	29.6573	34.4575		
P_{WG1} (MW)	23.6867	50.7413	49.5255	54.3755		
P_{TG3} (MW)	18.3428	16.7805	19.1227	20.5994		
P_{WG2} (MW)	16.9226	27.4983	33.0982	39.3083		
P_{SG1} (MW)	16.1253	18.9085	29.4924	33.8109		
V _{TG1} (pu)	1.0144	1.0488	1.0576	1.0589		
V _{TG2} (pu)	1.0040	1.0444	1.0480	1.0519		
V _{WG1} (pu)	0.9981	1.0369	1.0319	1.0365		
V _{TG3} (pu)	0.9908	1.0360	1.0371	1.0432		
V _{WG2} (pu)	1.0003	1.0748	1.0378	1.0666		
V _{SG1} (pu)	0.9957	1.0418	1.0307	1.0293		
Q _{C10} (MVar)	1.5008	1.4587	3.4604	1.9499		
Q _{C12} (MVar)	2.3222	0.0263	3.7348	3.3675		
Q _{C15} (MVar)	0.9766	2.0323	4.4984	2.1593		
Q _{C17} (MVar)	1.2045	2.6489	2.0928	1.7187		
Q _{C20} (MVar)	2.5313	1.7954	1.0430	2.1351		
Q _{C21} (MVar)	4.1492	4.7667	2.7054	3.0946		
Q _{C23} (MVar)	2.1710	2.8856	1.6427	1.4985		
Q _{C24} (MVar)	3.6903	0.9253	3.4766	1.9356		
Q _{C29} (MVar)	2.2945	2.7793	2.5635	0.9248		
T ₁₁ (pu)	0.9827	1.0439	0.9930	1.0014		
T ₁₂ (pu)	1.0065	0.9733	0.9794	1.0130		
T ₁₅ (pu)	1.0224	0.9740	0.9936	0.9962		
T ₃₆ (pu)	1.0061	0.9848	0.9786	0.9881		
TC_{sce} (\$/h)	429.9974	512.7754	580.1097	673.5231		
PL_{sce} (MW)	1.3858	1.1473	1.5306	2.0643		
E_{sce} (ton/h)	0.0992	0.1018	0.1004	0.1062		
<i>ETC</i> (\$/h)		548.	1304			
EPL (MW)	1.4615					
EE (ton/h)	0.1016					



FIGURE 13. Pareto front obtained by MOSGA for scenario 4.1.

4) SCENARIO 4: SIMULTANEOUS OPTIMIZATION OF TOTAL COST, ACTIVE POWER LOSS, AND EMISSION (UNCERTAIN LOAD DEMAND CASE STUDY)

This case study considers a practical variable situation of load demand. Normal PDF is utilized to reflect the uncertain nature of load [66]. Next, we apply the scenario-based method to implement the process of optimization at some discrete load levels. Fig. 12 portrays the load demand uncertainty in relation to the graphic form of normal distribution, in which the horizontal axis deputizes the percentage of network loading. The PDF parameters include a mean (μ_l) and a standard deviation (σ_l) of 70 and 10, respectively.



FIGURE 14. Pareto front obtained by MOSGA for scenario 4.2.



FIGURE 15. Pareto front obtained by MOSGA for scenario 4.3.



FIGURE 16. Pareto front obtained by MOSGA for scenario 4.4.

The color-filled areas manifest four different scenarios of network load demand (P_l) considered. For a certain scenario, the mean loading and its occurrence probability can be estimated by the following equations [66].

Probability of the i^{th} loading scenario is calculated by:

$$\delta_{sce,i} = \int_{P_{l,i}^{lo}}^{P_{l,i}^{up}} \frac{1}{\sigma_l \sqrt{2\pi}} \exp\left[-\frac{(P_l - \mu_l)}{2\sigma_l^2}\right] dP_l \qquad (58)$$

in which $P_{l,i}^{lo}$ and $P_{l,i}^{up}$ are the lower and upper bounds of the i^{th} loading scenario. Next, mean of the i^{th} loading

TABLE 13. Comparison of network performance profiles optimized by different algorithms for scenarios 4.1-4.4.

Algorithm	Scenario	TC_{sce} (\$/h)	PL_{sce} (MW)	E_{sce} (ton/h)	<i>ETC</i> (\$/h)	EPL (MW)	EE (ton/h)
MOSGA	Scenario 4.1	429.9974	1.3858	0.0992	548.1304	1.4615	0.1016
	Scenario 4.2	512.7754	1.1473	0.1018			
	Scenario 4.3	580.1097	1.5306	0.1004			
	Scenario 4.4	673.5231	2.0643	0.1062			
NSGA-II	Scenario 4.1	426.6767	1.2090	0.1005	536.9252	1.673	0.1019
	Scenario 4.2	491.6359	1.4759	0.1021			
	Scenario 4.3	571.5105	1.8390	0.1018			
	Scenario 4.4	670.1939	2.2040	0.1033			
MOALO	Scenario 4.1	434.2370	1.4254	0.0986	549.9583	1.5593	0.1024
	Scenario 4.2	511.8956	1.2447	0.1020			
	Scenario 4.3	584.7860	1.5873	0.0997			
	Scenario 4.4	672.6370	2.3093	0.1126			
MOGOA	Scenario 4.1	461.3624	1.0474	0.1001	555.102	1.4149	0.1025
	Scenario 4.2	508.7890	1.2727	0.1020			
	Scenario 4.3	591.1768	1.3538	0.1021			
	Scenario 4.4	670.8597	2.2197	0.1067			



FIGURE 17. Load bus voltage profiles for scenarios 4.1-4.4.

TABLE 14. Summary of the IEEE 57-bus system incorporating RESs.

System characteristics	Value	Detail
Bus	57	[59]
Branch	80	[59]
Thermal generator (TG1, TG2, TG3,	4	Buses: 1, 3, 8, and 12
TG4)		
Wind plant (WG1, WG2)	2	Buses: 6 and 9
Solar PV plant (SG1)	1	Buses: 2
Shunt VAR compensators	3	Buses: 18, 25 and 53
Transformers with tap changers	17	Branches: 19, 20, 31,
		35, 36, 37, 41, 46,
		54, 58, 59, 65, 66,
		71, 73, 76 and 80
Control variables	33	-

scenario is computed:

$$P_{l,i}^{\text{mean}} = \frac{1}{\delta_{sce,i}} \int_{P_{l,i}^{l_0}}^{P_{l,i}^{r_t}} \frac{1}{\sigma_l \sqrt{2\pi}} \exp\left[-\frac{(P_l - \mu_l)}{2\sigma_l^2}\right] dP_l \quad (59)$$

Table 11 provides information on the estimated means and probabilities for all loading scenarios. For a specific scenario, the demands in all nodes are adjusted by multiplying them by



FIGURE 18. The modified IEEE 57-bus system incorporating RES.

the percent loading, given the nominal loading of the network of 100%. At each scenario, the proposed MOSGA algorithm optimizes the multi-objective function as formulated in Eq. (50). Hence, the control variables of scheduled power from all generators, voltage magnitudes at generation nodes, transformer tap settings, and VAR compensators' output are optimized in each scenario of loading. Of note, the case study with the implementation of OPF at certain time intervals for various objectives optimization is similar to the practical operation condition of an electrical network.

Regarding the boundaries of control variables, they were set the same as in the fixed loading case. Simulation results related to various scenarios of loading of the 30-bus network are tabulated in Table 12 using MOSGA optimizer. Figs. 13-16 depict the Pareto optimal fronts found by the

 TABLE 15. Cost constant and emission constant of thermal generators for modified IEEE 57-bus system [2].

Generator	TG1	TG2	TG3	TG4
Bus	1	3	5	7
а	0	0	0	0
b	2	3	1	3.25
с	0.00375	0.025	0.0625	0.00834
α	0.04	0.04	0.05	0.06
β	-0.05	-0.05	-0.05	-0.05
γ	0.06	0.04	0.045	0.05
ω	0.00002	0.00001	0.00004	0.00001
μ	0.5	1.0	2.0	1.5

 TABLE 16. PDF parameters and cost constants for wind generators for modified IEEE 57-bus system [36].

Wind generator #	WG1 (bus 6)	WG2 (bus 9)
Number of wind turbines	50	40
Rated power (MW)	75	60
Weibull PDF parameters	c = 9, k = 2	c = 10, k = 2
Weibull mean	v = 7.976 m/s	v = 7.976 m/s
Direct cost constant (\$/MW)	3.2	3.5
Reserve cost constant (\$/MW)	3	3
Penalty cost constant (\$/MW)	6	6

 TABLE 17. PDF parameters and cost constants for solar PV generator for modified IEEE 57-bus system [36].

Solar PV #	SPV (bus 2)
Rated power (MW)	80
Lognormal PDF parameters	$\mu = 6, \sigma = 0.6$
Lognormal mean	$G = 483 \text{ W/m}^2$
Direct cost constant (\$/MW)	3.2
Reserve cost constant (\$/MW)	6
Penalty cost constant (\$/MW)	3

MOSGA for scenarios 4.1-4.4, respectively. As for each loading scenario, the optimized values of the total cost (TC_{sce}) , active power loss (PL_{sce}) , and emission (E_{sce}) are also reported in Table 12. Afterwards, the expected values of the total cost (ETC), active power loss (EPL), and emission (EE) over all scenarios can be computed according to Eqs. (60)-(62) [67]:

$$ETC = \sum_{sce=1}^{n_{sce}} \delta_{sce} \times TC_{sce}$$
(60)

$$EPL = \sum_{sce=1}^{n_{sce}} \delta_{sce} \times PL_{sce}$$
(61)

$$EE = \sum_{sce=1}^{n_{sce}} \delta_{sce} \times E_{sce}$$
(62)

in which, n_{sce} and δ_{sce} are the total numbers of load scenarios and the likelihood of a scenario, respectively (Table 11).

From the simulated outcomes in Tables 10 and 12, the total operational cost, power loss, and emission achieved by the uncertain loading case were justifiably lower than those achieved by the fixed loading case. The change in network loading resulted in a significant change in schedule power of generators, especially in solar PV and wind



FIGURE 19. Wind speed distribution for wind generators #1 at bus 6.



FIGURE 20. Wind speed distribution for wind generators #2 at bus 9.

power generators. Specifically, the optimal schedule power of generators increased when the load demand of network increased. To satisfy the increasing load demand, solar PV and wind generators were preferred to be scheduled. This may contribute to the notable improvement level of network performance profiles in examined discrete loading scenarios as well as of expected performance profiles over all scenarios. Compared with the fixed loading case, the uncertain loading case proposed a more flexible and effective scheduling of generators that led to lower total cost and power loss, and emission. Besides MOSGA, three multi-objective optimization algorithms including NSGA-II, MOALO, and MOGOA were also applied for results comparison in this case study. A comparison of results optimized by the four methods is summarized in Table 13. Clearly, MOSGA dominated MOALO and MOGOA in two out of three objectives in the best compromise solutions for all considered discrete loading scenarios. These helped MOSGA achieve lower three expected values (i.e., total cost, power loss and emission) over all scenarios than those of MOALO, whereas MOSGA achieved lower expected values of total cost and emission over all scenarios than those of MOGOA. Compared with NSGA-II, MOSGA obtained the best compromise solution with the better two out of three objectives for the second and third discrete loading scenarios while it excelled at only one objective for the first and fourth loading scenarios. Notably,

TABLE 18. Simulation results obtained by multi-objective algorithms for scenarios 5 and 6.

Parameters	Min	Max		Scen	ario 5			Scena	ario 6	
			MOSGA	NSGA-II	MOALO	MOGOA	MOSGA	NSGA-II	MOALO	MOGOA
P_{TG1} (MW)	0	575.88	131.4550	154.6992	120.1551	147.2304	178.1255	179.1888	206.9137	181.6055
P_{SG1} (MW)	0	80	80.0000	79.9244	80.0000	80.0000	80.0000	79.9951	79.9959	79.9575
P_{TG2} (MW)	40	140	52.0719	64.6718	50.1005	46.4158	80.4906	84.5556	84.6171	76.7879
P_{WG1} (MW)	0	150	150.0000	150.0000	123.2709	149.9913	150.0000	149.9982	150.0000	150.0000
P_{TG3} (MW)	100	550	326.6197	282.2865	360.4569	309.0656	332.8897	330.6756	365.2597	329.6130
P_{WG2} (MW)	0	120	119.8949	119.9736	120.0000	120.0000	120.0000	119.9969	120.0000	119.9994
P_{TG4} (MW)	100	410	402.6125	409.9684	409.2017	410.0000	323.8810	320.6863	263.0030	328.7181
V_{TG1} (pu)	0.95	1.1	1.0580	1.0364	1.0542	1.0719	1.0555	1.0417	1.0269	1.0808
V_{SG1} (pu)	0.95	1.1	1.0519	1.0325	1.0512	1.0671	1.0429	1.0376	1.0233	1.0730
V_{TG2} (pu)	0.95	1.1	1.0501	1.0270	1.0458	1.0593	1.0406	1.0327	1.0252	1.0517
V_{WG1} (pu)	0.95	1.1	1.0617	1.0341	1.0611	1.0562	1.0612	1.0406	1.0605	1.0665
V _{TG3} (pu)	0.95	1.1	1.0608	1.0299	1.0679	1.0324	1.0664	1.0407	1.0670	1.0682
V_{WG2} (pu)	0.95	1.1	1.0413	1.0160	1.0490	1.0222	1.0368	1.0185	1.0313	1.0427
V_{TG4} (pu)	0.95	1.1	1.0447	1.0306	1.0564	1.0374	1.0316	1.0218	1.0245	1.0401
Q_{C18} (MVar)	0	20	13.7042	10.9840	16.7307	4.9981	8.8921	10.5418	10.1151	2.7765
Q _{C25} (MVar)	0	20	14.5091	14.4975	19.9056	18.3832	12.6597	15.0978	11.8106	12.6753
Q _{C53} (MVar)	0	20	11.8367	12.7640	19.7594	15.1515	5.1954	12.3728	12.1821	4.8298
T ₁₉ (pu)	0.9	1.1	0.9334	0.9601	1.0582	1.0038	1.0526	0.9868	1.0511	0.9903
T_{20} (pu)	0.9	1.1	1.0863	0.9878	0.9818	0.9878	0.9886	0.9719	1.0930	1.0860
T_{31} (pu)	0.9	1.1	0.9777	1.0206	1.0531	1.0185	1.0254	1.0061	0.9471	0.9444
T ₃₅ (pu)	0.9	1.1	0.9993	1.0028	0.9641	1.0197	0.9445	1.0193	0.9559	0.9865
T_{36} (pu)	0.9	1.1	0.9721	1.0084	1.0930	1.0616	1.0738	0.9757	1.0106	0.9659
T ₃₇ (pu)	0.9	1.1	1.0100	1.0101	1.0124	1.0290	0.9864	1.0072	1.0198	1.0224
T ₄₁ (pu)	0.9	1.1	1.0071	0.9637	1.0112	0.9927	1.0048	0.9752	1.0238	0.9882
T ₄₆ (pu)	0.9	1.1	0.9783	0.9715	1.0264	0.9482	0.9625	0.9548	0.9930	0.9626
T ₅₄ (pu)	0.9	1.1	1.0438	0.9044	0.9619	1.0219	0.9635	0.9312	0.9848	0.9620
T ₅₈ (pu)	0.9	1.1	0.9950	0.9557	1.0232	0.9744	0.9693	0.9688	0.9452	0.9859
T ₅₉ (pu)	0.9	1.1	0.9860	0.9463	0.9837	1.0063	1.0034	0.9629	0.9659	0.9511
T ₆₅ (pu)	0.9	1.1	0.9743	0.9567	1.0603	0.9707	0.9885	0.9737	0.9467	0.9736
T ₆₆ (pu)	0.9	1.1	0.9841	0.9224	0.9636	0.9332	0.9406	0.9381	0.9456	1.0962
T ₇₁ (pu)	0.9	1.1	1.0086	0.9460	0.9856	0.9422	0.9551	0.9488	0.9651	0.9748
T ₇₃ (pu)	0.9	1.1	1.0086	0.9937	1.0703	0.9771	1.0125	0.9768	0.9733	1.0711
T ₇₆ (pu)	0.9	1.1	0.9640	0.9766	0.9366	1.0548	0.9374	0.9855	1.0577	0.9999
T ₈₀ (pu)	0.9	1.1	1.0225	0.9623	0.9906	0.9866	1.0133	0.9686	0.9788	1.0049
Total Cost (\$/h)	-	-	30212.29	30666.39	30944.81	30359.67	30575.27	30667.70	31043.30	30606.79
Real Power Loss (MW)	-	-	11.8540	10.7240	12.3851	11.9031	-	-	-	-
Emission (ton/h)	-	-	-	-	-	-	1.0043	0.9911	1.0220	1.0129



FIGURE 21. Solar irradiance distribution for solar PV at bus 2.



FIGURE 22. Available solar PV power (MW) at bus 2.

when considering the overall performance across all loading scenarios, MOSGA was superior in two out of three expected performance profiles over NSGA-II. Further, the voltage profiles at buses of the 30-bus network for all scenarios after the optimization by MOSGA are plotted in Fig. 17. The solutions obtained by MOSGA led to a significant improvement in bus voltage profiles for all loading scenarios while ensuring the permissible limits of bus voltage.

B. IEEE 57-BUS SYSTEM

Table 14 describes the 57-bus system incorporating RES; its topology diagram is shown in Fig. 18. Notably, locations of

TABLE 19.	Simulation result	s obtained by	/ multi-objective	algorithms for
scenario 7.				

TABLE 20. Simulation results obtained by proposed MOSGA algorithm for loading scenarios 8.1-8.4.

$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Parameters	MOSGA	NSGA-II	MOALO	MOGOA
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	P _{TG1} (MW)	159.9512	176.2043	202.7236	173.3578
$\begin{array}{llllllllllllllllllllllllllllllllllll$	P_{SG1} (MW)	78.3546	79.8647	80.0000	69.7586
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	P_{TG2} (MW)	67.7556	79.1197	68.3884	89.5941
$\begin{array}{llllllllllllllllllllllllllllllllllll$	P_{WG1} (MW)	150.0000	149.8122	150.0000	149.6149
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	P_{TG3} (MW)	291.8846	288.4601	302.2815	286.7908
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	P_{WG2} (MW)	120.0000	119.8832	120.0000	119.9460
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	P_{TG4} (MW)	394.6979	369.3791	340.9147	374.2587
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{TG1} (pu)	1.0504	1.0498	1.0615	1.0434
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{SG1} (pu)	1.0389	1.0430	1.0512	1.0389
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{TG2} (pu)	1.0386	1.0376	1.0507	1.0393
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{WG1} (pu)	1.0552	1.0423	1.0643	1.0403
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{TG3} (pu)	1.0545	1.0434	1.0656	1.0416
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{WG2} (pu)	1.0313	1.0243	1.0507	1.0232
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	V _{TG4} (pu)	1.0335	1.0347	1.0644	1.0342
$\begin{array}{c cccc} Q_{C25} \left(\text{MVar} \right) & 17.5603 & 13.8036 & 14.3377 & 19.2641 \\ Q_{C33} \left(\text{MVar} \right) & 10.7133 & 11.4029 & 14.0523 & 8.0197 \\ T_{19} \left(\text{pu} \right) & 1.0494 & 0.9766 & 1.0025 & 1.0288 \\ T_{20} \left(\text{pu} \right) & 1.0168 & 0.9764 & 1.0215 & 0.9582 \\ T_{31} \left(\text{pu} \right) & 1.0030 & 0.9946 & 1.0236 & 0.9931 \\ T_{35} \left(\text{pu} \right) & 1.0201 & 0.9906 & 0.9970 & 0.9616 \\ T_{37} \left(\text{pu} \right) & 1.0019 & 0.9760 & 0.9996 & 1.0234 \\ T_{46} \left(\text{pu} \right) & 1.0019 & 0.9760 & 0.9996 & 1.0234 \\ T_{46} \left(\text{pu} \right) & 1.0003 & 0.9469 & 1.0091 & 0.9876 \\ T_{54} \left(\text{pu} \right) & 1.0077 & 0.9521 & 1.0527 & 0.9615 \\ T_{58} \left(\text{pu} \right) & 1.0170 & 0.9725 & 0.9800 & 0.9721 \\ T_{59} \left(\text{pu} \right) & 0.9523 & 0.9689 & 0.9888 & 0.9961 \\ T_{65} \left(\text{pu} \right) & 0.9548 & 0.9726 & 0.9926 & 1.0461 \\ T_{66} \left(\text{pu} \right) & 0.9538 & 0.9595 & 0.9726 & 0.9440 \\ T_{73} \left(\text{pu} \right) & 0.9564 & 1.0018 & 0.9803 & 1.0093 \\ T_{75} \left(\text{pu} \right) & 0.9666 & 0.9752 & 0.9908 & 1.0093 \\ T_{60} \left(\text{pu} \right) & 0.96648 & 0.9726 & 0.9926 & 1.0461 \\ T_{66} \left(\text{pu} \right) & 0.9538 & 0.9595 & 0.9726 & 0.9440 \\ T_{73} \left(\text{pu} \right) & 0.9564 & 1.0018 & 0.9803 & 1.0093 \\ T_{76} \left(\text{pu} \right) & 0.9663.58 & 30823.51 & 30763.29 & 31421.10 \\ \text{Real Power Loss} & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ \left(\text{MW} \right) \end{array}$	Q _{C18} (MVar)	15.1187	10.3863	15.4906	4.9504
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Q _{C25} (MVar)	17.5603	13.8036	14.3377	19.2641
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Q _{C53} (MVar)	10.7133	11.4029	14.0523	8.0197
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₁₉ (pu)	1.0494	0.9766	1.0025	1.0288
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₂₀ (pu)	1.0168	0.9764	1.0215	0.9582
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₃₁ (pu)	1.0030	0.9946	1.0236	0.9931
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₃₅ (pu)	0.9896	1.0048	1.0107	1.0481
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₃₆ (pu)	1.0201	0.9906	0.9970	0.9616
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₃₇ (pu)	1.0079	1.0022	0.9908	1.0436
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₄₁ (pu)	1.0019	0.9760	0.9996	1.0234
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₄₆ (pu)	1.0003	0.9469	1.0091	0.9876
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	T ₅₄ (pu)	1.0077	0.9521	1.0527	0.9615
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	T ₅₈ (pu)	1.0170	0.9725	0.9800	0.9721
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	T ₅₉ (pu)	0.9523	0.9689	0.9888	0.9961
$\begin{array}{cccccccc} T_{66} \left(pu \right) & 0.9743 & 0.9397 & 0.9896 & 1.0058 \\ T_{71} \left(pu \right) & 0.9538 & 0.9595 & 0.9726 & 0.9440 \\ T_{73} \left(pu \right) & 0.9564 & 1.0018 & 0.9803 & 1.0093 \\ T_{76} \left(pu \right) & 0.9495 & 0.9907 & 0.9968 & 0.9025 \\ T_{80} \left(pu \right) & 0.9866 & 0.9752 & 0.9992 & 1.0140 \\ Total Cost (\$/h) & 30663.58 & 30823.51 & 30763.29 & 31421.10 \\ Real Power Loss & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ (MW) \\ Emission (ton/h) & 1.0846 & 1.0168 & 1.0091 & 1.0261 \\ \end{array}$	T ₆₅ (pu)	0.9648	0.9726	0.9926	1.0461
$\begin{array}{cccccccc} T_{71} \left(pu \right) & 0.9538 & 0.9595 & 0.9726 & 0.9440 \\ T_{73} \left(pu \right) & 0.9564 & 1.0018 & 0.9803 & 1.0093 \\ T_{76} \left(pu \right) & 0.9495 & 0.9907 & 0.9968 & 0.9025 \\ T_{80} \left(pu \right) & 0.9866 & 0.9752 & 0.9992 & 1.0140 \\ Total Cost \left(\$ h \right) & 30663.58 & 30823.51 & 30763.29 & 31421.10 \\ Real Power Loss & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ (MW) \\ Emission \left(ton/h \right) & 1.0846 & 1.0168 & 1.0091 & 1.0261 \\ \end{array}$	T ₆₆ (pu)	0.9743	0.9397	0.9896	1.0058
$\begin{array}{cccccc} T_{73} \left(pu \right) & 0.9564 & 1.0018 & 0.9803 & 1.0093 \\ T_{76} \left(pu \right) & 0.9495 & 0.9907 & 0.9968 & 0.9025 \\ T_{80} \left(pu \right) & 0.9866 & 0.9752 & 0.9992 & 1.0140 \\ Total Cost (\$/h) & 30663.58 & 30823.51 & 30763.29 & 31421.10 \\ Real Power Loss & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ (MW) \\ Emission \left(ton/h \right) & 1.0846 & 1.0168 & 1.0091 & 1.0261 \\ \end{array}$	T ₇₁ (pu)	0.9538	0.9595	0.9726	0.9440
$\begin{array}{ccccc} T_{76} \left(pu \right) & 0.9495 & 0.9907 & 0.9968 & 0.9025 \\ T_{80} \left(pu \right) & 0.9866 & 0.9752 & 0.9992 & 1.0140 \\ Total Cost (\$/h) & 30663.58 & 30823.51 & 30763.29 & 31421.10 \\ Real Power Loss & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ (MW) & & & & \\ Emission \left(ton/h \right) & 1.0846 & 1.0168 & 1.0091 & 1.0261 \\ \end{array}$	T ₇₃ (pu)	0.9564	1.0018	0.9803	1.0093
$\begin{array}{cccccc} T_{80} \left(pu \right) & 0.9866 & 0.9752 & 0.9992 & 1.0140 \\ Total Cost \left(\$/h \right) & 30663.58 & 30823.51 & 30763.29 & 31421.10 \\ Real Power Loss & 11.8439 & 11.9233 & 13.5082 & 12.5209 \\ (MW) & & & & & \\ Emission \left(ton/h \right) & 1.0846 & 1.0168 & 1.0091 & 1.0261 \\ \end{array}$	T ₇₆ (pu)	0.9495	0.9907	0.9968	0.9025
Total Cost (\$/h) 30663.58 30823.51 30763.29 31421.10 Real Power Loss 11.8439 11.9233 13.5082 12.5209 (MW) Emission (ton/h) 1.0846 1.0168 1.0091 1.0261	T ₈₀ (pu)	0.9866	0.9752	0.9992	1.0140
Real Power Loss 11.8439 11.9233 13.5082 12.5209 (MW) 1.0168 1.0091 1.0261	Total Cost (\$/h)	30663.58	30823.51	30763.29	31421.10
(MW) Emission (ton/h) 1.0846 1.0168 1.0091 1.0261	Real Power Loss	11.8439	11.9233	13.5082	12.5209
Emission (ton/h) 1.0846 1.0168 1.0091 1.0261	(MW)				
	Emission (ton/h)	1.0846	1.0168	1.0091	1.0261

Parameters	Scenario	Scenario	Scenario	Scenario
	4.2	4.3	4.3	4.4
P_{TG1} (MW)	63.5990	60.3554	99.6512	118.7943
P_{SG1} (MW)	73.4479	79.3816	79.3954	68.2492
P_{TG2} (MW)	44.8917	47.9778	65.3644	63.3125
P_{WG1} (MW)	147.8180	148.8233	146.1072	150.0000
P_{TG3} (MW)	100.0000	174.2209	179.3368	246.8164
P_{WG2} (MW)	112.3398	120.0000	118.6825	119.7239
P_{TG4} (MW)	148.7603	194.9880	252.5502	308.9685
V _{TG1} (pu)	1.0096	1.0190	1.0341	1.0392
V _{SG1} (pu)	1.0100	1.0188	1.0392	1.0348
V _{TG2} (pu)	1.0226	1.0235	1.0412	1.0264
V _{WG1} (pu)	1.0367	1.0337	1.0468	1.0380
V _{TG3} (pu)	1.0279	1.0216	1.0403	1.0293
V _{WG2} (pu)	1.0255	1.0095	1.0195	1.0177
V _{TG4} (pu)	1.0257	0.9957	1.0102	1.0260
Q _{C18} (MVar)	2.3330	8.4819	11.6109	11.4148
Q _{C25} (MVar)	11.4329	7.5471	11.2996	12.6569
Q _{C53} (MVar)	13.4633	8.8970	8.9685	6.1451
T ₁₉ (pu)	1.0644	1.0120	1.0878	1.0535
T_{20} (pu)	0.9527	1.0142	1.0732	1.0077
T_{31} (pu)	1.0069	1.0298	0.9518	1.0137
T ₃₅ (pu)	1.0951	1.0629	0.9886	1.0543
T ₃₆ (pu)	0.9833	1.0329	1.0388	1.0124
T ₃₇ (pu)	1.0261	1.0405	1.0335	0.9637
T ₄₁ (pu)	0.9886	0.9756	1.0261	0.9779
T ₄₆ (pu)	0.9270	0.9318	0.9893	0.9507
T ₅₄ (pu)	0.9851	1.0135	1.0147	1.0213
T ₅₈ (pu)	0.9885	1.0026	1.0156	0.9629
T ₅₉ (pu)	0.9894	1.0040	0.9949	0.9511
T ₆₅ (pu)	0.9581	0.9656	0.9546	0.9643
T ₆₆ (pu)	0.9715	0.9311	0.9368	0.9774
T ₇₁ (pu)	0.9639	1.0253	1.0123	0.9355
T ₇₃ (pu)	0.9273	1.0009	0.9375	1.0294
T ₇₆ (pu)	1.0044	0.9924	0.9866	0.9025
T ₈₀ (pu)	0.9905	0.9840	0.9761	0.9584
TC_{sce} (\$/h)	11100.56	14634.46	18837.96	23534.84
PL_{sce} (MW)	6.0607	7.7067	8.0082	9.5417
E_{sce} (ton/h)	0.1998	0.3105	0.4338	0.6799
<i>ETC</i> (\$/h)		16920).7749	
EPL (MW)		7.8	396	
EE (ton/h)		0.3	936	

RESs were selected the same as the references [36] by replacing some conventional generators with respective renewable generators. Table 15 details the fuel cost and emission factors of thermal generators. Tables 16-17 provide the PDF parameters and cost constants of wind and solar PV generators. Figs. 19-20 depict wind frequency distributions for two wind farms after Monte-Carlo simulations with 8000 scenarios obtained for each. Fig. 21 displays the frequency distribution of solar irradiance for a solar PV plant. Stochastic power output from the solar PV plant is depicted in Fig. 22. For the initial scenario, the system has a total cost of 51795.8561 \$/h, a real power loss of 27.5627 MW, and an emission of 2.6474 ton/h.

1) SCENARIO 5: SIMULTANEOUS OPTIMIZATION OF TOTAL COST AND ACTIVE POWER LOSS

In this scenario, the MOSGA method simultaneously optimized total cost and real power loss objectives. Fig. 23 presents the Pareto optimal front obtained by the MOSGA for scenario 5. Similar to scenario 1, scenario 5 also shows the conflicting relationship between the total cost and real power loss. Table 18 shows the optimization values of the best compromise solution from the MOSGA, NSGA-II, MOALO, and MOGOA. The best compromise solution from MOSGA provided a total cost of 30212.29 \$/h (i.e., 41.6705 % total cost reduction), along with a real power loss of 11.8540 MW (i.e., 56.9926 % real power loss reduction) compared to their initial scenario values. From Table 18, MOSGA (30212.29 \$/h and 11.8540 MW) obtained lower total cost and real power loss values than MOALO (30944.81 \$/h and 12.3851 MW) and MOGOA (30359.67 \$/h and 11.9031 MW), which indicated that MOSGA dominated those of MOALO and MOGOA for best compromise solution. In addition, MOSGA outperformed NSGA-II in one out of two objectives optimized. Hence, MOSGA was able to highly compete with MOALO, MOGOA, and NSGA-II in finding the high-quality best compromise solution.

TABLE 21. Comparison of network performance profiles optimized by different algorithms for scenarios 8.1-8.4.

Algorithm	Scenario	TC_{sce} (\$/h)	PL_{sce} (MW)	E_{sce} (ton/h)	<i>ETC</i> (\$/h)	EPL (MW)	EE (ton/h)
MOSGA	Scenario 4.1	11100.56	6.0607	0.1998	16920.7749	7.8396	0.3936
	Scenario 4.2	14634.46	7.7067	0.3105			
	Scenario 4.3	18837.96	8.0082	0.4338			
	Scenario 4.4	23534.84	9.5417	0.6799			
NSGA-II	Scenario 4.1	11324.33	4.4141	0.2240	17203.9955	5.9416	0.4327
	Scenario 4.2	15259.48	4.9911	0.3629			
	Scenario 4.3	18870.84	6.6900	0.4807			
	Scenario 4.4	23680.82	7.9036	0.6883			
MOALO	Scenario 4.1	12267.25	5.0256	0.2359	17616.4555	6.8869	0.4306
	Scenario 4.2	15692.63	6.2541	0.3445			
	Scenario 4.3	19209.82	7.5076	0.4678			
	Scenario 4.4	23676.35	8.7744	0.7306			
MOGOA	Scenario 4.1	10992.92	6.6343	0.2289	17094.5259	7.3934	0.4235
	Scenario 4.2	15146.37	5.9813	0.3600			
	Scenario 4.3	18659.91	8.5108	0.4467			
	Scenario 4.4	24019.32	8.7860	0.7047			



FIGURE 23. Pareto front obtained by MOSGA for scenario 5.



FIGURE 24. Pareto front obtained by MOSGA for scenario 6.

2) SCENARIO 6: SIMULTANEOUS OPTIMIZATION OF TOTAL COST AND EMISSION

The total cost and emission were simultaneously optimized using the MOSGA in this scenario. The Pareto optimal front yielded by MOSGA is described in Fig. 24. Table 18 shows the simulation outputs from four algorithms for scenario 6. The total cost and emission were reduced to 30575.27 \$/h and 1.0043 ton/h for the best compromise solution, corresponding



FIGURE 25. Pareto front obtained by MOSGA for scenario 7.



FIGURE 26. Load bus voltage profiles for scenarios 5, 6, and 7.

to 40.9697% and 62.0652% reductions. Therefore, both objectives were significantly improved upon compared to the initial scenario. In this case, MOSGA obtained better results than MOALO and MOGOA for the total cost and emission objectives. Meanwhile, the optimal solution of MOSGA was better than NSGA-II in the objective of total cost only. Hence, MOSGA had great potential in finding better compromise solution.



FIGURE 27. Pareto front obtained by MOSGA for scenario 8.1.



FIGURE 28. Pareto front obtained by MOSGA for scenario 8.2.



FIGURE 29. Pareto front obtained by MOSGA for scenario 8.3.

3) SCENARIO 7: SIMULTANEOUS OPTIMIZATION OF TOTAL COST, ACTIVE POWER LOSS, AND EMISSION

In scenario 7, the MOSGA simultaneously minimized the total cost, real power loss, and emission. Fig. 25 portrays the Pareto optimal front found by MOSGA. Table 19 gives the simulation results generated by the MOSGA, NSGA-II, MOALO, and MOGOA for the best compromise solution in this scenario. For MOSGA, the total cost, real power loss, and emission were reduced to 30663.58 \$/h, 11.8439 MW, and 1.0846 ton/h, respectively. Hence, MOSGA found a total cost reduction of 40.7992% and a power loss reduction of 57.0292%, along with a reduction of 59.0327% in emission for the best compromise solution. Although MOSGA



FIGURE 30. Pareto front obtained by MOSGA for scenario 8.4.



FIGURE 31. Load bus voltage profiles for scenarios 8.1-8.4.

was not able to directly dominate NSGA-II, MOALO, and MOGOA, MOSGA obtained better total cost and real power loss values than the others in the best compromise solution. Hence, MOSGA was more advantageous in handling the MOOPF-WS problem with the large-scale system.

It can be seen that the implementation of the MOSGA for the MOOPF-WS problem offers a notable improvement for all three objective function values. Fig. 26 depicts the voltage profiles of load buses for scenarios 5-7, which shows that the voltage limits were all within acceptable limitations.

4) SCENARIO 8: SIMULTANEOUS OPTIMIZATION OF TOTAL COST, ACTIVE POWER LOSS, AND EMISSION (UNCERTAIN LOAD DEMAND CASE STUDY)

Similarly, experiment outcomes for the uncertain load case of the 57-bus network obtained by MOSGA are presented in Table 20. Figs. 27-30 show the Pareto optimal fronts found by the MOSGA for scenarios 8.1-8.4, respectively. Comparing the results between Table 19 and Table 20 shows the uncertain case of load demand offered justifiably lower network performance profiles of total cost, power loss, and emission than those of the fixed loading case. Again, the scheduling of generators from the uncertain load case was more suitable with the variable load demand. Solar PV and wind power generators were preferable to schedule due to their positive impacts in reducing network operation cost,

TABLE 22. Comparisons of multi-objective methods using Δ indicator for scenarios 1-8.

Scenario	MOSGA		NSC	iA-II	MO	ALO	MOGOA	
-	Average	Standard	Average	Standard	Average	Standard	Average	Standard
		deviation		deviation		deviation		deviation
Scenario 1	0.5737	0.0393	1.0522	0.0364	1.0906	0.0681	1.4364	0.2089
Scenario 2	0.8593	0.0532	1.1079	0.0665	1.1744	0.0889	1.5547	0.1583
Scenario 3	0.8727	0.0396	1.0265	0.0346	1.1240	0.0927	1.3775	0.1769
Scenario 4.1	0.7385	0.0304	1.0206	0.0185	1.0153	0.0119	1.3410	0.1204
Scenario 4.2	0.8976	0.0313	1.0140	0.0167	1.0159	0.0189	1.3993	0.1720
Scenario 4.3	0.8997	0.0294	1.0137	0.0118	1.0302	0.0345	1.4540	0.1873
Scenario 4.4	0.8798	0.0297	1.0347	0.0231	1.0597	0.0341	1.4717	0.2144
Scenario 5	0.9323	0.0312	1.0332	0.0468	1.0021	0.0026	1.0470	0.0414
Scenario 6	0.6778	0.0415	1.0734	0.0432	1.0259	0.0115	1.3240	0.1990
Scenario 7	0.8394	0.0554	1.0297	0.0260	1.0085	0.0050	1.1165	0.0825
Scenario 8.1	0.8124	0.0690	1.0176	0.0208	1.0065	0.0038	1.1095	0.1484
Scenario 8.2	0.8049	0.0524	1.0136	0.0113	1.0140	0.0112	1.1346	0.1415
Scenario 8.3	0.7954	0.0585	1.0208	0.0144	1.0122	0.0067	1.1828	0.1178
Scenario 8.4	0.8343	0.0573	1.0128	0.0121	1.0077	0.0041	1.1875	0.1500

TABLE 23. Comparisons of multi-objective methods using HV indicator for scenarios 1-8.

Scenario	MOSGA		NSC	JA-II	MO	ALO	MOGOA	
_	Average	Standard	Average	Standard	Average	Standard	Average	Standard
		deviation		deviation		deviation		deviation
Scenario 1	0.7487	0.0042	0.5719	0.0260	0.4898	0.0769	0.6688	0.0234
Scenario 2	0.9285	0.0015	0.6898	0.0715	0.6784	0.1312	0.8933	0.0112
Scenario 3	0.6707	0.0115	0.5173	0.0253	0.4341	0.1108	0.6017	0.0286
Scenario 4.1	0.4852	0.0308	0.2807	0.0574	0.1963	0.0548	0.4295	0.0609
Scenario 4.2	0.7100	0.0289	0.4725	0.0523	0.3052	0.1192	0.5976	0.0700
Scenario 4.3	0.7644	0.0199	0.5679	0.0268	0.4105	0.1030	0.6538	0.0515
Scenario 4.4	0.7061	0.0160	0.5603	0.0158	0.3780	0.1092	0.6129	0.0371
Scenario 5	0.8336	0.0273	0.7691	0.0473	0.2869	0.1344	0.5787	0.1922
Scenario 6	0.8046	0.0083	0.6200	0.0316	0.2761	0.1254	0.6327	0.2138
Scenario 7	0.4460	0.0698	0.4167	0.0840	0.1768	0.0527	0.2700	0.0893
Scenario 8.1	0.4814	0.0435	0.4078	0.0572	0.1692	0.0504	0.2651	0.0872
Scenario 8.2	0.5189	0.0443	0.4481	0.0457	0.1702	0.0741	0.3119	0.1337
Scenario 8.3	0.5333	0.0466	0.4626	0.0480	0.1920	0.0806	0.3072	0.1136
Scenario 8.4	0.5016	0.0467	0.4304	0.0578	0.1845	0.0563	0.2683	0.1197



FIGURE 32. Box plots of \triangle metric of multi-objective algorithms for scenarios 1-8.

power loss, and emission. To confirm the effectiveness of MOSGA, the obtained best compromise solutions related to four scenarios of loading were compared with those of

NSGA-II, MOALO, and MOGOA (Table 21). Obviously, MOSGA showed its domination in two of three objectives when compared with NSGA-II, MOALO, and MOGOA for



FIGURE 33. Box plots of HV metric of multi-objective algorithms for scenarios 1-8.

all discrete loading scenarios. As a result, MOSGA acquired lower three expected values of the total cost, power loss, and emission over all scenarios than MOALO. In comparison with NSGA-II and MOGOA, MOSGA obtained lower two expected values of total cost and emission over all scenarios. Moreover, for all scenarios, bus voltage profiles of the 57-bus network were significantly enhanced and completely satisfied with the limits after the optimization by MOSGA (Fig. 31).

C. STATISTICAL COMPARISONS AND ANALYSIS

This section demonstrates the comparisons among MOSGA, NSGA-II, MOALO, and MOGOA based on Δ and HV metrics. The initial parameters of all algorithms were retained at the same values as in the preceding section.

1) SPREAD (Δ) METRIC

Table 22 shows statistical results attained by the MOSGA, NSGA-II, MOALO, and MOGOA for the Δ indicator for scenarios 1-8. Boxplots of the statistical analysis for Δ metric are depicted in Fig. 32. As can be seen, the MOSGA provided the best solutions for the Δ metric, with the best distribution for Pareto optimal fronts among the four methods for all scenarios. Hence, the proposed MOSGA showed superiority over the others for scenarios 1-8.

2) HYPERVOLUME (HV) METRIC

Table 23 gives the optimization results of four methods for the HV metric for scenarios 1-8. Figs. 33 shows the boxplots of the HV metric for scenarios 1-8. From Table 23 and Fig. 33, MOSGA was better than the other methods for the average value of the HV metric, which indicated that solutions yielded by the MOSGA had the best convergence and diversity.

TABLE 24. Comparisons of multi-objective methods in terms of computational time in seconds.

Scenario	MOSGA	NSGA-II	MOALO	MOGOA
Scenario 1	148.3135	355.9984	176.8823	345.9891
Scenario 2	147.4339	347.0797	175.2286	342.0755
Scenario 3	144.4651	340.8604	173.9698	346.2734
Scenario 4.1	145.9328	356.9026	174.0651	350.2344
Scenario 4.2	145.6260	342.7177	174.1245	350.2964
Scenario 4.3	138.3401	328.2177	178.9667	353.3510
Scenario 4.4	139.3932	330.6724	184.9313	359.9552
Scenario 5	278.8974	629.9411	346.8786	728.2135
Scenario 6	273.5693	626.9396	352.6557	753.5177
Scenario 7	276.1229	644.7474	347.4177	742.8672
Scenario 8.1	274.2589	635.2438	357.7776	746.1943
Scenario 8.2	274.5833	649.5932	357.5052	756.2760
Scenario 8.3	271.5552	628.1484	351.5760	793.7339
Scenario 8.4	277.1661	633.0042	347.6344	739.2438

Therefore, MOSGA found a better quality Pareto optimal front in comparison to NSGA-II, MOALO, and MOGOA. As seen in Figs. 33, MOSGA was also superior to the other methods in terms of stability and robustness.

3) COMPUTATIONAL TIME

In this study, MOSGA, NSGA-II, MOALO and MOGOA were executed on a 2.6 GHz 4-core computer with 16 GB of RAM. Table 24 presents the average computational times of MOSGA, NSGA-II, MOALO and MOGOA for all scenarios. The computational times of MOSGA were significantly faster than those of NSGA-II, MOALO, and MOGOA for all case studies. Hence, MOSGA outperformed the other methods for the solution quality, robustness, and computational times in all cases.

VI. CONCLUSION

This study suggested a new MOSGA for the MOOPF-WS problem, in which thermal, wind and solar PV generators were integrated into the grid. Different PDFs were used to model uncertainties relative to RES and load demand as well. The MOSGA integrated fast non-dominated sorting and crowding distance methods in defining the non-dominated ranks and densities of the solutions obtained. Furthermore, the Pareto archive selection strategy was applied for the distribution maintenance of the non-dominated solutions. The MOSGA was implemented successfully to the adapted 30-bus and 57-bus systems incorporating RES where various combinations of objective functions and different loading scenarios were examined. The findings showed that system performance is significantly improved for the best compromise solution in both fixed and variable loading scenarios. Notably, in scenario 7, the total cost, real power loss, and emission were decreased by 40.7992%, 57.029%, and 59.0327%, respectively, in comparison with the initial scenario. For the variable loading case, the expected performance profiles over all discrete loading scenarios were significantly lower than the respective ones found by the fixed loading case. This proved that solving OPF in the variable load situation offered more flexible and effective scheduling of generators than in the fixed load situation, resulting in lower expected performance profiles. Moreover, the performance of the MOSGA was compared with NSGA-II, MOALO, and MOGOA using Δ and HV indicators. In all cases, the MOSGA showed superiority in convergence and diversity of Pareto optimum solutions compared with the other three methods. As a result, the MOSGA could detect a broader range of non-dominated solutions for all objective functions. Furthermore, when comparing with significant research in the literature, MOSGA obtained better solution quality in all the comparable cases. These revealed the strengths of the MOSGA and validated its potential in dealing with the MOOPF-WS problem. In future studies, the MOOPF-WS problem can be expanded in some specific aspects such as connection consideration of smallhydro source, electric vehicles, and flexible AC transmission system (FACTS) devices.

ACKNOWLEDGMENT

The authors acknowledge the support of the Ministry of Higher Education (MOHE), Malaysia to provide financial assistance under Fundamental Research Grant Scheme (FRGS) Reference number: FRGS/1/2018/ TK07/UTP/02/3 and Universiti Teknologi PETRONAS to provide the laboratory facilities to conduct this research work.

REFERENCES

- J. Carpentier, "Contribution a l'Etude du Dispatching Economique," Bull. de la Societe Francaise des Electriciens, vol. 3, pp. 431–474, 1962.
- [2] A. E. Chaib, H. R. E. H. Bouchekara, R. Mehasni, and M. A. Abido, "Optimal power flow with emission and non-smooth cost functions using backtracking search optimization algorithm," *Int. J. Elect. Power Energy Syst.*, vol. 81, pp. 64–77, Oct. 2016, doi: 10.1016/ j.ijepes.2016.02.004.

- [3] R. C. Burchett, H. H. Happ, D. R. Vierath, and K. A. Wirgau, "Developments in optimal power flow," *IEEE Trans. Power App. Syst.*, vol. PAS-101, no. 2, pp. 406–414, Feb. 1982, doi: 10.1109/TPAS. 1982.317121.
- [4] K. Zehar and S. Sayah, "Optimal power flow with environmental constraint using a fast successive linear programming algorithm: Application to the Algerian power system," *Energy Convers. Manage.*, vol. 49, no. 11, pp. 3362–3366, Nov. 2008, doi: 10.1016/j.enconman.2007.10.033.
- [5] G. P. Granelli and M. Montagna, "Security-constrained economic dispatch using dual quadratic programming," *Electr. Power Syst. Res.*, vol. 56, no. 1, pp. 71–80, 2000, doi: 10.1016/S0378-7796(00)00097-3.
- [6] O. Crisan and M. A. Mohtadi, "Efficient identification of binding inequality constraints in the optimal power flow Newton approach," *IEE Proc. C Gener, Transmiss. Distrib.*, vol. 139, no. 5, pp. 365–370, Sep. 1992, doi: 10.1049/ip-c.1992.0053.
- [7] J. A. Momoh and J. Z. Zhu, "Improved interior point method for OPF problems," *IEEE Trans. Power Syst.*, vol. 14, no. 3, pp. 1114–1120, Aug. 1999, doi: 10.1109/59.780938.
- [8] S. Frank, I. Steponavice, and S. Rebennack, "Optimal power flow: A bibliographic survey I," *Energy Syst.*, vol. 3, no. 3, pp. 221–258, Sep. 2012, doi: 10.1007/s12667-012-0056-y.
- [9] W. Warid, H. Hizam, N. Mariun, and N. I. A. Wahab, "A novel quasioppositional modified Jaya algorithm for multi-objective optimal power flow solution," *Appl. Soft Comput.*, vol. 65, pp. 360–373, Apr. 2018, doi: 10.1016/j.asoc.2018.01.039.
- [10] H. R. E. H. Bouchekara, A. E. Chaib, M. A. Abido, and R. A. El-Schiemy, "Optimal power flow using an improved colliding bodies optimization algorithm," *Appl. Soft Comput.*, vol. 42, pp. 119–131, May 2016, doi: 10.1016/j.asoc.2016.01.041.
- [11] A. E. Chaib, H. R. E. H. Bouchekara, R. Mehasni, and M. A. Abido, "Optimal power flow with emission and non-smooth cost functions using backtracking search optimization algorithm," *Int. J. Elect. Power Energy Syst.*, vol. 81, pp. 64–77, Oct. 2016, doi: 10.1016/j.ijepes.2016.02.004.
- [12] K. Abaci and V. Yamacli, "Differential search algorithm for solving multiobjective optimal power flow problem," *Int. J. Electr. Power Energy Syst.*, vol. 79, pp. 1–10, Jul. 2016, doi: 10.1016/j.ijepes.2015.12.021.
- [13] A.-A. A. Mohamed, Y. S. Mohamed, A. A. M. El-Gaafary, and A. M. Hemeida, "Optimal power flow using moth swarm algorithm," *Electr. Power Syst. Res.*, vol. 142, pp. 190–206, Jan. 2017, doi: 10.1016/j. epsr.2016.09.025.
- [14] P. P. Biswas, P. N. Suganthan, R. Mallipeddi, and G. A. J. Amaratunga, "Optimal power flow solutions using differential evolution algorithm integrated with effective constraint handling techniques," *Eng. Appl. Artif. Intell.*, vol. 68, pp. 81–100, Feb. 2018, doi: 10.1016/ j.engappai.2017.10.019.
- [15] A. M. Shaheen, S. M. Farrag, and R. A. El-Schiemy, "MOPF solution methodology," *IET Gener., Transmiss. Distrib.*, vol. 11, no. 2, pp. 570–581, 2016, doi: 10.1049/iet-gtd.2016.1379.
- [16] H. Pulluri, R. Naresh, and V. Sharma, "An enhanced self-adaptive differential evolution based solution methodology for multiobjective optimal power flow," *Appl. Soft Comput.*, vol. 54, pp. 229–245, May 2017, doi: 10.1016/j.asoc.2017.01.030.
- [17] X. Yuan, B. Zhang, P. Wang, J. Liang, Y. Yuan, Y. Huang, and X. Lei, "Multi-objective optimal power flow based on improved strength Pareto evolutionary algorithm," *Energy*, vol. 122, pp. 70–82, Mar. 2017, doi: 10.1016/j.energy.2017.01.071.
- [18] G. Chen, X. Yi, Z. Zhang, and H. Wang, "Applications of multi-objective dimension-based firefly algorithm to optimize the power losses, emission, and cost in power systems," *Appl. Soft Comput.*, vol. 68, pp. 322–342, Jul. 2018, doi: 10.1016/j.asoc.2018.04.006.
- [19] H. Tazvinga, B. Zhu, and X. Xia, "Optimal power flow management for distributed energy resources with batteries," *Energy Convers. Manage.*, vol. 102, pp. 104–110, Sep. 2015, doi: 10.1016/j.enconman.2015.01.015.
- [20] A. Panda, M. Tripathy, A. K. Barisal, and T. Prakash, "A modified bacteria foraging based optimal power flow framework for hydro-thermalwind generation system in the presence of STATCOM," *Energy*, vol. 124, pp. 720–740, Apr. 2017, doi: 10.1016/j.energy.2017.02.090.
- [21] R. A. Jabr and B. C. Pal, "Intermittent wind generation in optimal power flow dispatching," *IET Generation, Transmiss. Distrib.*, vol. 3, no. 1, pp. 66–74, Jan. 2009, doi: 10.1049/IET-GTD:20080273.
- [22] S. Mishra, Y. Mishra, and S. Vignesh, "Security constrained economic dispatch considering wind energy conversion systems," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2011, pp. 1–8, doi: 10.1109/ PES.2011.6039544.

- [23] L. Shi, C. Wang, L. Yao, Y. Ni, and M. Bazargan, "Optimal power flow solution incorporating wind power," *IEEE Syst. J.*, vol. 6, no. 2, pp. 233–241, Jun. 2012, doi: 10.1109/JSYST.2011.2162896.
- [24] A. Panda and M. Tripathy, "Optimal power flow solution of wind integrated power system using modified bacteria foraging algorithm," *Int. J. Elect. Power Energy Syst.*, vol. 54, pp. 306–314, Jan. 2014, doi: 10.1016/j.ijepes.2013.07.018.
- [25] A. Panda and M. Tripathy, "Security constrained optimal power flow solution of wind-thermal generation system using modified bacteria foraging algorithm," *Energy*, vol. 93, pp. 816–827, Dec. 2015, doi: 10.1016/j.energy.2015.09.083.
- [26] S. S. Reddy, "Optimal power flow with renewable energy resources including storage," *Electr. Eng.*, vol. 99, no. 2, pp. 685–695, Jun. 2017, doi: 10.1007/s00202-016-0402-5.
- [27] K. Teeparthi and D. M. V. Kumar, "Security-constrained optimal power flow with wind and thermal power generators using fuzzy adaptive artificial physics optimization algorithm," *Neural Comput. Appl.*, vol. 29, no. 3, pp. 855–871, Feb. 2018, doi: 10.1007/s00521-016-2476-4.
- [28] R. Roy and H. T. Jadhov, "Optimal power flow solution of power system incorporating stochastic wind power using Gbest guided artificial bee colony algorithm," *Int. J. Elect. Power Energy Syst.*, vol. 64, pp. 562–578, Jan. 2015, doi: 10.1016/j.ijepes.2014.07.010.
- [29] A. Panda and M. Tripathy, "Solution of wind integrated thermal generation system for environmental optimal power flow using hybrid algorithm," *J. Electr. Syst. Inf. Technol.*, vol. 3, no. 2, pp. 151–160, Sep. 2016, doi: 10.1016/j.jesit.2016.01.004.
- [30] P. P. Biswas, P. N. Suganthan, and G. A. J. Amaratunga, "Optimal power flow solutions incorporating stochastic wind and solar power," *Energy Convers. Manage.*, vol. 148, pp. 1194–1207, Sep. 2017, doi: 10.1016/j.enconman.2017.06.071.
- [31] M.-R. Chen, G.-Q. Zeng, and K.-D. Lu, "Constrained multi-objective population extremal optimization based economic-emission dispatch incorporating renewable energy resources," *Renew. Energy*, vol. 143, pp. 277–294, Dec. 2019, doi: 10.1016/j.renene.2019.05.024.
- [32] U. Guvenc, S. Duman, H. T. Kahraman, S. Aras, and M. Katı, "Fitnessdistance balance based adaptive guided differential evolution algorithm for security-constrained optimal power flow problem incorporating renewable energy sources," *Appl. Soft Comput.*, vol. 108, Sep. 2021, Art. no. 107421, doi: 10.1016/j.asoc.2021.107421.
- [33] N. Karthik, A. K. Parvathy, R. Arul, and K. Padmanathan, "Multi-objective optimal power flow using a new heuristic optimization algorithm with the incorporation of renewable energy sources," *Int. J. Energy Environ. Eng.*, vol. 12, no. 4, pp. 641–678, Dec. 2021, doi: 10.1007/s40095-021-00397-x.
- [34] I. U. Khan, N. Javaid, K. A. A. Gamage, C. J. Taylor, S. Baig, and X. Ma, "Heuristic algorithm based optimal power flow model incorporating stochastic renewable energy sources," *IEEE Access*, vol. 8, pp. 148622–148643, 2020, doi: 10.1109/ACCESS.2020.3015473.
- [35] M. H. Sulaiman and Z. Mustaffa, "Solving optimal power flow problem with stochastic wind–solar–small hydro power using barnacles mating optimizer," *Control Eng. Pract.*, vol. 106, Jan. 2021, Art. no. 104672, doi: 10.1016/j.conengprac.2020.104672.
- [36] S. Duman, S. Rivera, J. Li, and L. Wu, "Optimal power flow of power systems with controllable wind-photovoltaic energy systems via differential evolutionary particle swarm optimization," *Int. Trans. Electr. Energy Syst.*, vol. 30, no. 4, Apr. 2020, Art. no. e12270, doi: 10.1002/2050-7038.12270.
- [37] E. Kaymaz, S. Duman, and U. Guvenc, "Optimal power flow solution with stochastic wind power using the Lévy coyote optimization algorithm," *Neural Comput. Appl.*, vol. 33, no. 12, pp. 6775–6804, Jun. 2021, doi: 10.1007/s00521-020-05455-9.
- [38] A. R. Abul'Wafa, "Optimization of economic/emission load dispatch for hybrid generating systems using controlled elitist NSGA-II," *Electric Power Syst. Res.*, vol. 105, pp. 142–151, Dec. 2013, doi: 10.1016/j. epsr.2013.07.006.
- [39] B. Y. Qu, J. J. Liang, Y. S. Zhu, Z. Y. Wang, and P. N. Suganthan, "Economic emission dispatch problems with stochastic wind power using summation based multi-objective evolutionary algorithm," *Inf. Sci.*, vol. 351, pp. 48–66, Jul. 2016, doi: 10.1016/j.ins.2016.01.081.
- [40] T. C. Bora, V. C. Mariani, and L. D. S. Coelho, "Multi-objective optimization of the environmental-economic dispatch with reinforcement learning based on non-dominated sorting genetic algorithm," *Appl. Thermal Eng.*, vol. 146, pp. 688–700, Jan. 2019, doi: 10.1016/j. applthermaleng.2018.10.020.

- [41] Y. Zhu, J. Wang, and B. Qu, "Multi-objective economic emission dispatch considering wind power using evolutionary algorithm based on decomposition," *Int. J. Electr. Power Energy Syst.*, vol. 63, pp. 434–445, Dec. 2014, doi: 10.1016/j.ijepes.2014.06.027.
- [42] R. Azizipanah-Abarghooee, T. Niknam, A. Roosta, A. R. Malekpour, and M. Zare, "Probabilistic multiobjective wind-thermal economic emission dispatch based on point estimated method," *Energy*, vol. 37, no. 1, pp. 322–335, Jan. 2012, doi: 10.1016/j.energy.2011.11.023.
- [43] N. A. Khan, A. B. Awan, A. Mahmood, S. Razzaq, A. Zafar, and G. A. S. Sidhu, "Combined emission economic dispatch of power system including solar photo voltaic generation," *Energy Convers. Manage.*, vol. 92, pp. 82–91, Mar. 2015, doi: 10.1016/j.enconman.2014.12.029.
- [44] M. S. Gonçalves, R. H. Lopez, and L. F. F. Miguel, "Search group algorithm: A new Metaheuristic method for the optimization of truss structures," *Comput. Struct.*, vol. 153, pp. 165–184, Jun. 2015, doi: 10.1016/j.compstruc.2015.03.003.
- [45] G. Acampora, D. Caruso, A. Vaccaro, and A. Vitiello, "A search group algorithm for optimal voltage regulation in power systems," in *Proc. IEEE Congr. Evol. Comput. (CEC)*, Jul. 2016, pp. 3662–3669, doi: 10.1109/ CEC.2016.7744253.
- [46] U. K. Rout, R. K. Sahu, and S. Panda, "Gravitational search algorithm based automatic generation control for interconnected power system," in *Proc. Int. Conf. Circuits, Power Comput. Technol. (ICCPCT)*, Mar. 2013, pp. 557–568, doi: 10.1109/iccpct.2013.6528998.
- [47] S. F. H. Noorbin and A. Alfi, "Adaptive parameter control of search group algorithm using fuzzy logic applied to networked control systems," *Soft Comput.*, vol. 22, no. 23, pp. 7939–7960, Dec. 2018, doi: 10.1007/s00500-017-2742-0.
- [48] F. Carraro, R. H. Lopez, and L. F. F. Miguel, "Optimum design of planar steel frames using the search group algorithm," *J. Brazilian Soc. Mech. Sci. Eng.*, vol. 39, no. 4, pp. 1405–1418, Apr. 2017, doi: 10.1007/s40430-016-0628-1.
- [49] R. L. Pedro, J. Demarche, L. F. F. Miguel, and R. H. Lopez, "An efficient approach for the optimization of simply supported steel-concrete composite I-girder bridges," *Adv. Eng. Softw.*, vol. 112, pp. 31–45, Oct. 2017, doi: 10.1016/j.advengsoft.2017.06.009.
- [50] P. P. Biswas, P. N. Suganthan, B. Y. Qu, and G. A. J. Amaratunga, "Multiobjective economic-environmental power dispatch with stochastic windsolar-small hydro power," *Energy*, vol. 150, pp. 1039–1057, May 2018, doi: 10.1016/j.energy.2018.03.002.
- [51] H. M. Dubey, M. Pandit, and B. K. Panigrahi, "Hybrid flower pollination algorithm with time-varying fuzzy selection mechanism for wind integrated multi-objective dynamic economic dispatch," *Renew. Energy*, vol. 83, pp. 188–202, Nov. 2015, doi: 10.1016/j.renene.2015.04.034.
- [52] J. Hetzer, D. C. Yu, and K. Bhattarai, "An economic dispatch model incorporating wind power," *IEEE Trans. Energy Convers.*, vol. 23, no. 2, pp. 603–611, Jun. 2008, doi: 10.1109/TEC.2007.914171.
- [53] X. Liu and W. Xu, "Economic load dispatch constrained by wind power availability: A here-and-now approach," *IEEE Trans. Sustain. Energy*, vol. 1, no. 1, pp. 2–9, Apr. 2010, doi: 10.1109/TSTE.2010.2044817.
- [54] X. Liu, "Economic load dispatch constrained by wind power availability: A wait-and-see approach," *IEEE Trans. Smart Grid*, vol. 1, no. 3, pp. 347–355, Dec. 2010, doi: 10.1109/TSG.2010.2057458.
- [55] T. P. Chang, "Investigation on frequency distribution of global radiation using different probability density functions," *Int. J. Appl. Sci. Eng.*, vol. 8, no. 2, pp. 99–107, Sep. 2010, doi: 10.6703/IJASE.2010.8(2).99.
- [56] S. S. Reddy, P. R. Bijwe, and A. R. Abhyankar, "Real-time economic dispatch considering renewable power generation variability and uncertainty over scheduling period," *IEEE Syst. J.*, vol. 9, no. 4, pp. 1440–1451, Dec. 2015, doi: 10.1109/JSYST.2014.2325967.
- [57] D. E. Goldberg, Genetic Algorithms in Search, Optimization and Machine Learning, 1st ed. Reading, MA, USA: Addison-Wesley, 1989.
- [58] K. Deb, A. Pratap, S. Agarwal, and T. Meyarivan, "A fast and elitist multiobjective genetic algorithm: NSGA-II," *IEEE Trans. Evol. Comput.*, vol. 6, no. 2, pp. 182–197, Jan. 2002, doi: 10.1109/4235.996017.
- [59] R. D. Zimmerman and C. E. Murillo-Sanchez, MATPOWER-Manual-4.1.Pdf. Cornell University, Ithaca, NY, USA: Power Systems Engineering Research Center, 2011. Accessed: Sep. 19, 2020. [Online]. Available: https://matpower.org/docs/MATPOWER-manual-4.1.pdf
- [60] W. Sheng, K.-Y. Liu, Y. Liu, X. Meng, and Y. Li, "Optimal placement and sizing of distributed generation via an improved nondominated sorting genetic algorithm II," *IEEE Trans. Power Del.*, vol. 30, no. 2, pp. 569–578, Apr. 2015, doi: 10.1109/TPWRD.2014.2325938.

- [61] Y.-N. Wang, L.-H. Wu, and X.-F. Yuan, "Multi-objective self-adaptive differential evolution with elitist archive and crowding entropy-based diversity measure," *Soft Comput.*, vol. 14, no. 3, p. 193, Jan. 2009, doi: 10.1007/s00500-008-0394-9.
- [62] K. Deb, Multi-Objective Optimization Using Evolutionary Algorithms. Hoboken, NJ, USA: Wiley, 2001.
- [63] O. Alsac and B. Stott, "Optimal load flow with steady-state security," *IEEE Trans. Power App. Syst.*, vol. PAS-93, no. 3, pp. 745–751, May 1974, doi: 10.1109/TPAS.1974.293972.
- [64] S. Mirjalili, P. Jangir, and S. Saremi, "Multi-objective ant lion optimizer: A multi-objective optimization algorithm for solving engineering problems," *Appl. Intell.*, vol. 46, no. 1, pp. 79–95, Jan. 2017, doi: 10.1007/s10489-016-0825-8.
- [65] S. Z. Mirjalili, S. Mirjalili, S. Saremi, H. Faris, and I. Aljarah, "Grasshopper optimization algorithm for multi-objective optimization problems," *Appl. Intell.*, vol. 48, no. 4, pp. 805–820, 2018, doi: 10.1007/s10489-017-1019-8.
- [66] S. M. Mohseni-Bonab, A. Rabiee, and B. Mohammadi-Ivatloo, "Voltage stability constrained multi-objective optimal reactive power dispatch under load and wind power uncertainties: A stochastic approach," *Renew. Energy*, vol. 85, pp. 598–609, Jan. 2016, doi: 10.1016/j. renene.2015.07.021.
- [67] P. P. Biswas, P. N. Suganthan, R. Mallipeddi, and G. A. J. Amaratunga, "Optimal reactive power dispatch with uncertainties in load demand and renewable energy sources adopting scenario-based approach," *Appl. Soft Comput.*, vol. 75, pp. 616–632, Feb. 2019, doi: 10.1016/j. asoc.2018.11.042.



NURSYARIZAL MOHD NOR received the M.Sc. degree in electrical power engineering from The University of Manchester Institute of Science and Technology (UMIST), U.K., in 2001, and the Ph.D. degree in electrical engineering from Universiti Teknologi PETRONAS (UTP), Malaysia, in 2009. His areas of specialization are 'analysis and optimization of large-scale power systems' and 'state estimation'. He has several publications at his credit. His research interests include power

system state estimation, power system analysis, renewable energy, and electrical machine.



IRRAIVAN ELAMVAZUTHI (Member, IEEE) received the Ph.D. degree from the Department of Automatic Control & Systems Engineering, The University of Sheffield, U.K., in 2002. He is currently an Associate Professor at the Department of Electrical and Electronic Engineering, Universiti Teknologi PETRONAS (UTP), Malaysia. His research interests include control, robotics, mechatronics, power systems, and bio-medical applications. He is also the Chair of the IEEE Robotics

and Automation Society (Malaysia Chapter).



TRUONG HOANG BAO HUY received the B.Eng. degree in electrical and electronics engineering from the Ho Chi Minh City University of Technology (HCMUT), VNU-HCM, Vietnam, in 2017, and the M.S. degree in electrical and electronics engineering from Universiti Teknologi PETRONAS (UTP), Malaysia, in 2020. He is currently a Research Officer with the Institute of Engineering and Technology, Thu Dau Mot University, Binh Duong, Vietnam. His research inter-

ests include renewable energy power generation, artificial intelligence-based algorithms, and their application in optimization problems.



TAIB IBRAHIM was born in Kedah, Malaysia, in 1972. He received the B.Eng. degree (Hons.) in electrical and electronics engineering from Coventry University, U.K., in 1996, the M.Sc. degree in electrical power engineering from the University of Strathclyde, U.K., in 2000, and the Ph.D. degree in electrical machine design from The University of Sheffield, U.K., in 2009. His employment experience includes Airod (M) Sdn Bhd and Universiti Teknologi PETRONAS (UTP). He is currently the

Leader for power and energy cluster and the Co-Leader for mission-oriented research (energy) with the UTP. His research interests include motion control to electromagnetic devices and their associated drives.



TRI PHUOC NGUYEN received the B.Eng. degree in electrical engineering from Can Tho University, Can Tho City, Vietnam, in 2014, and the M.Eng. degree in electrical engineering from the Ho Chi Minh City University of Technology, Ho Chi Minh City, Vietnam, in 2017. His research interest includes applications of metaheuristic methods to optimization of distribution systems.



DIEU NGOC VO received the B.Eng. and M.Eng. degrees in electrical engineering from the Ho Chi Minh City University of Technology (HCMUT), VNU-HCM, Ho Chi Minh City, Vietnam, in 1995 and 2000, respectively, and the D.Eng. degree in energy from the Asian Institute of Technology (AIT), Pathum Thani, Thailand, in 2007. He is currently an Associate Professor with the Department of Power Systems, Faculty of Electrical and Electronic Engineering, HCMUT. His

research interests include applications of AI in power system optimization, power system operation and control, power system analysis, and power systems under deregulation.